

Temi di discussione

(Working Papers)

Fiscal buffers, private debt and recession: the good, the bad and the ugly

by Nicoletta Batini, Giovanni Melina and Stefania Villa







Temi di discussione

(Working Papers)

Fiscal buffers, private debt and recession: the good, the bad and the ugly

by Nicoletta Batini, Giovanni Melina and Stefania Villa

Number 1186 - July 2018

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Antonio Bassanetti, Marco Casiraghi, Emanuele Ciani, Vincenzo Cuciniello, Nicola Curci, Davide Delle Monache, Giuseppe Ilardi, Andrea Linarello, Juho Taneli Makinen, Valentina Michelangeli, Valerio Nispi Landi, Marianna Riggi, Lucia Paola Maria Rizzica, Massimiliano Stacchini. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

FISCAL BUFFERS, PRIVATE DEBT AND RECESSION: THE GOOD, THE BAD AND THE UGLY

by Nicoletta Batini*, Giovanni Melina* and Stefania Villa**

Abstract

Focusing on Euro-Area countries, we show empirically that higher private debt leads to deeper recessions while higher public debt does not, unless its level is especially high. We then build a general equilibrium model that replicates these dynamics and use it to design a policy that can mitigate the recessionary consequences of private deleveraging. In the model, in the aftermath of financial shocks, recessions are milder and public debt is more contained when the government lends directly to those households and firms that face binding borrowing constraints. As a consequence, large fiscal buffers are critical to enhance macroeconomic resilience to financial shocks.

JEL Classification: E44, E62, H63.

Keywords: private debt, public debt, financial crisis, financial shocks, borrowing constraints, fiscal limits.

Contents

1. Introduction	5
2. Stylized facts	7
3. Model	
4. Model calibration and validation	
5. A deleveraging episode	
6. Conclusion	
References	
Appendix A: countries in panel regressions and descriptive statistics	44
Appendix B: private debt and GDP growth at different frequencies	44
Appendix C: equilibrium conditions	46
Appendix D: robustness checks to selected parameter values	50
Appendix E: the role of fiscal buffers in the case of targeted intervention	51

^{*} International Monetary Fund. E-mail: nbatini@imf.org and gmelina@imf.org. ** Fellow at the Bank of Italy, Directorate General for Economics, Statistics and Research. E-mail: stefania.villa@esterni.bancaditalia.it.

1 Introduction¹

The global financial crisis followed an extraordinary upward swing in the leverage cycle in a number of advanced countries (Geanakoplos et al., 2012). When the bubble burst, the massive debt accumulation in the private sector sparked a typical debt deflation (Fisher, 1933; Minsky, 1982) that propelled the ratio of public debt-to-GDP very rapidly. This reflected, on one side, the recession-induced decline in government revenues and prices, including those of assets; and, on the other side, governments directly taking over private debt gone sour.

Spurred by such economic developments, late empirical studies have started to focus increasingly more on the relationship between private debt and the macroeconomy. Part of this literature documents the links between rapid credit growth – especially credit to households – and financial crises in the advanced world (Glick and Lansing, 2010; Dell'Ariccia et al., 2012; IMF, 2012; Schularick and Taylor, 2012; Taylor, 2012; Jordà et al., 2013; Mian and Sufi, 2014; Jordà et al., 2014). The key messages from this body of research are that excessive credit growth predicts financial crises and that, conditional on having a recession, stronger credit growth predicts deeper recessions. Mian et al. (2017) take these results a level further, finding unconditional negative correlations between household debt changes and future growth in a panel of advanced and emerging market economies. In addition, they demonstrate that rises in public debt are not associated with lower future output growth.

Theoretical economic modeling has flanked the empirical research, at least up to a certain point. Building upon the modern macroeconomic model-based literature on collateral and leverage cycles (pioneered by Bernanke et al., 1999; Kiyotaki and Moore, 1997; and Iacoviello, 2005) a number of recent papers in macro-finance have

¹The views expressed in this paper are those of the authors and do not necessarily represent those of Bank of Italy, the International Monetary Fund or IMF policy. We are grateful to Antonio Bassanetti, Olivier Blanchard, Giovanni Callegari, Alessandro Cantelmo, Efrem Castelnuovo, Pietro Catte, Jacopo Cimadomo, Riccardo Cristadoro, Lorenzo Forni, Vitor Gaspar, Matteo Iacoviello, Alejandro Justiniano, Prakash Loungani, Daniela Marconi, Valerio Nispi Landi, Francesco Nucci, Maurice Obstfeld; seminar participants at Banca d'Italia, Federal Reserve Board and IMF; conference participants at the 2015 ECB Conference on "Debt overhang, macroeconomic adjustment and EMU economic governance", the IMAEF 2016 Conference, the 12th Dynare Conference, the 2017 ICMAIF conference, the "Finance and Economic Growth in the Aftermath of the Crisis" conference at the University of Milan, the "Secular Stagnation and Financial Cycles" conference at Banca d'Italia, and the 2017 "Macroeconomic Policy Meetings" at the Melbourne Institute for useful comments. All remaining errors are ours.

focused on how to reproduce mechanisms through which excessive indebtedness in the private sector can harm the economy (e.g. Eggertsson and Krugman, 2012; Farhi and Werning, 2016; Korinek and Simsek, 2016; Guerrieri and Lorenzoni, 2017; Martin and Philippon, 2017).

However, there are important gaps in the literature. First, none of these contributions studies the macro-financial interlinkages between private and public balance sheets that so distinctly characterized both the evolution and the recovery phases of the recent crisis in the advanced world. Second, models featuring a fully-fledged public sector facing fiscal limits (such as Cantore et al., 2018) do not factor in the role of the government as a lender of last resort during protracted phases of financial stress. And third, research so far – starting with Gertler and Karadi (2011) – has focused exclusively on the impact of central bank lending to banks during a financial crisis, abstracting from other equally effective policy tools, like the possibility – observed during the financial crisis – that the government lends directly to financially-constrained agents.

In this paper we move forward by making positive and normative additions to the existing literature to fill these gaps. Focusing on the Euro Area, an economy in which the loop between private and public debt has played a particularly important role during the Great Recession, we begin by highlighting three stylized facts. The first two reaffirm those unveiled by Mian et al. (2017) for a broader set of countries, i.e. a negative correlation between increases in private debt and future levels of output, and a lack of correlation between a rise in public debt and the severity of future recessions. However, we show that in those Euro-Area countries where the level of public debt is especially elevated, increases in public debt lead more severe recessions.

We then build a dynamic stochastic general equilibrium (DSGE) model calibrated on the Euro Area that reproduces these stylized facts and can help us both examine the macroeconomic effects of a financial-crisis-style event leading to private deleveraging and conduct policy experiments.² The basic structure of the model embeds Iacoviello (2005)'s features of borrowing constraints in the housing market within a New-Keynesian setting. The model is enriched with a fiscal bloc that tracks changes in the stock of government debt and captures the links between this and the sovereign risk premium, as in Cantore et al. (2018). In this set up, through the

²We do not, however, attempt to explicitly model the global financial crisis, nor the sovereign debt crisis. Modeling the systemic aspect of the financial risk is beyond the scope of this paper.

financial accelerator, private deleveraging depresses output and prices, which in turn depresses government revenues, accelerating the accumulation of public debt. Crucially, we assume that the government can alleviate private borrowing constraints via targeted lending operations (dubbed henceforth "targeted interventions"), which impinge on debt-to-GDP dynamics by affecting output and deficits. By virtue of these features, the model can mimic well the key links between debt and output that characterized the recent financial crisis in the Euro Area.

Model simulations suggest that, running into a financial shock with high private debt is as, or possibly more, worrisome than confronting such a shock with high public debt. They also suggest that government loans to borrowing-constrained agents during deleveraging phases mitigates the adverse recessionary effects of financial shocks on output, whilst offering net gains for the government medium-run fiscal position. At the same time, these results do not call for limitless financial assistance to the private sector. In fact, we unveil clear trade-offs between costs and benefits of targeted interventions. For countries closer to their fiscal limit, the benefits are more muted because they are undone by subsequent tax hikes necessary to preserve fiscal sustainability. It follows that one of the key benefits of having fiscal buffers is the greater macroeconomic resilience to financial shocks.

The remainder of the paper is organized as follows. Section 2 presents stylized facts on the links between debt, both private and public, and future output in the Euro Area. Section 3 describes the model. Section 4 discusses the model calibration and validation. Section 5 studies a deleveraging episode and government targeted intervention. Finally, Section 6 concludes. Technical details and robustness checks are appended to the paper.

2 Stylized facts

To pin down the relationship between private debt, public debt and future output for the Euro Area, in this section we fit a panel regression to an unbalanced annual panel dataset encompassing eleven countries and stretching from 1960 to 2014 (Appendix A.1 reports summary statistics and the sample period available for private and public debt for each country).³

 $^{^{3}}$ Data on private debt are taken from the BIS dataset "Long series on total credit to the private non-financial sector". Private debt is computed as the sum of total credit (financed by domestic and foreign banks as well as nonbank financial institutions) to households and non-

We use the same dynamic structure as in the baseline specification proposed by Mian et al. (2017) who, without trying to prove causality, study correlations between past private and public debt, and future GDP in a panel of thirty advanced and emerging market economies. Our analysis differs from theirs along three dimensions: (i) we focus exclusively on the eleven Euro Area countries; (ii) for consistency with our DSGE model, we study the interlinkages between the cyclical components (HPfiltered) of private and public debt-to-GDP ratios and output, as opposed to debt and output growth rates (as we show below, our results hold also using growth rates); and, finally, (iii) we repeat the econometric analysis for two subsamples of countries grouped by levels of public debt.

Both the empirical analysis reported in this section and the model-based simulations reported in the remainder of the paper focus on business cycle considerations. This is why the main objects of the study are short-term fluctuations of macro variables. Appendix B investigates the correlations between private debt and GDP growth at different frequencies, putting these into the context of the broader growth literature.

Our panel regression equation is as follows:

$$\hat{y}_{it+3} = \alpha_i + \beta_{prd} \left(\frac{P\hat{R}D}{Y}\right)_{i,t} + \beta_{pud} \left(\frac{P\hat{U}D}{Y}\right)_{i,t} + u_{i,t},\tag{1}$$

where the dependent variable is future, three year-out detrended real GDP, $\hat{y}_{i,t+3}$, in line with previous studies;⁴ while the two regressors are the detrended total private debt-to-GDP ratio, $\left(\frac{P\hat{R}D}{Y}\right)_{i,t}$, and the detrended public debt-to-GDP ratio; *i* indexes a generic country; α_i indicates country-specific constants; β_{prd} and β_{pud} are our coefficients of interest, and $u_{i,t}$ is the error term.

Table 1, column 1, shows that the cyclical component of private debt is inversely related to that of future output. Likewise, column 2 shows that the cyclical component of the public debt-to-GDP ratio is not correlated to that of output three years

financial corporations. For public debt we use data from the IMF World Economic Outlook dataset. Real output is taken from IMF International Financial Statistics.

⁴The rationale for the choice of the three-year horizon is based on: (i) the empirical evidence provided by Mian et al. (2017), who justify the three-year horizon to examine the effect of credit expansion in the households' sector on output in a VAR setting; (ii) findings of optimal lag in terms of strongest predictive power found by Baron and Xiong (2017) who, more similarly to us, use total bank credit to GDP instead of household vs. non-financial corporations' debt separately; and (iii) work by Dell'Ariccia et al. (2012) who show that the median bank credit boom lasts three years.

	Dependent variable: \hat{y}_{it+3}			
	(1)	(2)	(3)	(4)
$\overline{\left(\frac{P\hat{R}D}{Y}\right)_{it}}$	-0.133***	-0.132***	-0.130***	-0.151***
	(0.018)	(0.019)	(0.019)	(0.050)
$\left(\frac{P\hat{U}D}{Y}\right)_{it}$		0.002	0.069**	-0.127***
		(0.027)	(0.032)	(0.045)
"Low" public debt	\checkmark	\checkmark	\checkmark	
"High" public debt	\checkmark	\checkmark		\checkmark
R^2	0.157	0.171	0.249	0.154
Country fixed effects	\checkmark	\checkmark	\checkmark	\checkmark
Observations	317	257	154	103

 Table 1: Cyclical Fluctuations of Private and Public Debt and Subsequent Cyclical

 Fluctuations of Real GDP

Notes: Estimates are obtained via panel regressions of deviations of real GDP from HP(100) trend in t + 3 on the detrended private and public debt to GDP ratio in t. All specifications include country fixed effects. *,**,**** denote significance at the 0.1, 0.05, 0.01 level, respectively.

later when we condition on the level of total private debt. These results for the Euro Area confirm more general results obtained by Mian et al. (2017) for a wider sample of countries and with a different transformation of the data.

In columns 3 and 4 we try to uncover possible nonlinearities in the relationship depending on the average level of public debt over GDP. To do so, we divide our sample into two groups: a "low" public debt group, including only countries with public debt below or equal to the Euro Area median value of 63.2 percent of GDP⁵ (Finland, France, Germany, Ireland, Netherlands and Spain), and a "high" public debt group, including only countries with public debt above the Euro Area median (Austria, Belgium, Greece, Italy and Portugal). Results indicate that in both groups private debt remains consistently negatively associated with future GDP. Instead, in those countries where the level of public debt is "low", the public debt ratio is positively correlated with subsequent GDP (column 3). In contrast, in those countries where the level of public debt is "high", the public debt ratio is negatively correlated with subsequent GDP (column 4).

As a robustness check, we re-run the regressions using the same data transformation as Mian et al. (2017). Specifically, the dependent variable is future output

⁵The median value is computed as the median of the mean values of public debt-to-GDP for each EA country over the sample period.

growth over three years, $\Delta_3 y_{i,t+3}$, while the two regressors are the change in total private debt-to-GDP ratio in the previous three years, $\Delta_3 \left(\frac{PRD}{Y}\right)_{i,t-1}$, and the change in public debt-to-GDP ratio, again in the previous three years, $\Delta_3 \left(\frac{PUD}{Y}\right)_{i,t-1}$. Columns 1 and 2 in Table 2 confirm, for the Euro Area sample, that also under this specification changes in private debt are inversely related to output growth in the near future, while changes in the ratio of public debt-to-GDP are insignificantly related to future output growth. The non-linear effect of the relationship with respect to the average level of public debt re-emerges as well in this specification, with "low" public debt countries exhibiting a positive correlation between public debt and growth, and "high" public debt countries exhibiting an inverse relation.

Our results are reminiscent of those of Schularick and Taylor (2012), Taylor (2012) and Jordà et al. (2013) who find that the bigger the credit boom going bust, the more painful the subsequent recession, a pain that is proportionally worse for countries also saddled with an adverse fiscal position (Jordà et al., 2014). One important corollary that can be drawn from these stylized facts is that countries with larger fiscal buffers are better positioned to weather a financial crisis, both because they have the room needed to allow automatic stabilizers to work fully and because they can, in case, offer stabilizing fiscal support to credit-constrained agents. To formally analyze these issues we use the DSGE model outlined in the following section.

3 Model

We build a model featuring financial frictions in the Kiyotaki and Moore (1997)-Iacoviello (2005) closed-economy tradition. We innovate upon this framework by extending the basic structure to account for fiscal policy, government debt and the sovereign risk premium. These additional features enable us to spell out clearly how the interlinkages between private and public debt play out.

The economy is populated by patient households (lenders), impatient households (borrowers), entrepreneurs, the government and the central bank. Patient households work, consume, buy housing, invest in riskless private bonds and in government bond holdings. Impatient households work, consume, and borrow subject to collateral constraints. Entrepreneurs also borrow subject to a collateral constraint and produce in monopolistic competition. The government finances its expenditures by

		Dependent variable: $\Delta_3 y_{i,t+3}$			
	(1)	(2)	(3)	(4)	
$\Delta_3 \left(\frac{PRD}{Y}\right)_{i,t-1}$	-0.239***	-0.219***	-0.195***	-0.362***	
	(0.024)	(0.029)	(0.029)	(0.068)	
$\Delta_3 \left(\frac{PUD}{Y}\right)_{i,t-1}$		-0.016	0.064^{*}	-0.214***	
-,		(0.036)	(0.038)	(0.067)	
"Low" public debt	\checkmark	\checkmark	\checkmark		
"High" public debt	\checkmark	\checkmark		\checkmark	
R^2	0.261	0.232	0.273	0.346	
Country fixed effects	\checkmark	\checkmark	\checkmark	\checkmark	
Observations	283	218	131	87	

Table 2: Private and Public Debt and Subsequent Real GDP Growth

Notes: Estimates are obtained via panel regressions of real GDP growth from t to t + 3 on either the change in private and public debt to GDP from t - 4 to t - 1 or the level of private and public debt in t - 1. All specifications include country fixed effects. *,**,*** denote significance at the 0.1, 0.05, 0.01 level, respectively.

raising a mix of lump-sum and distortionary taxes and by issuing government bonds. Holding government debt is subject to sovereign default risk and the fiscal limit is calibrated in line with the Greek default case as in Cantore et al. (2018), among others. The government also has a role as a lender of last resort. So beyond being tasked – as usual – with the provision of public goods financed through taxation and with standard tools to smooth the economic cycle, it may also lend to creditconstrained agents in the aftermath of financial shocks. Monetary policy follows a Taylor-type rule, while the fiscal rule is set such that government expenditures and taxes react to stabilize public debt compatibly with the government's fiscal limits. Finally, to keep the model simple, but without loss of generality, we do not include banks.⁶

It is important to note a few, but important definitional conventions in the paper. By *leverage cycle* we denote an increase (decrease) in private indebtedness caused

⁶Financial intermediaries are essentially intermediaries between the ultimate lenders and borrowers. Their debt reduction does not influence the assessment of sustainability of the debt burden to the economy, which is the focus of this work. Including banks would add financial frictions, and under certain modeling assumptions, could be set in a way as to magnify leverage cycles by allowing a greater mismatch between debt maturities and risk between ultimate borrowers and lenders. If at all, this would buttress the economic forces driving our results, not lessen them. Therefore, if anything, our policy implications are starker in that we underestimate the financial accelerator effect.

by a loosening (tightening) of borrowing constraints when the collateral of both impatient households and entrepreneurs appreciates (depreciates) in value. With the term *deleveraging*, we indicate a reduction in liabilities achieved through cuts to borrowing to keep these in line with the value of the collateral. A *crisis* occurs when a drop in the value of the collateral reduces the availability of credit to borrow out of future income. In the paper, *targeted intervention* refers to credit extended by the government to financially-constrained agents to alleviate their borrowing constraints.

The model simulations reported throughout the paper are produced with perturbation methods, taking a first-order linear approximation of the model's equilibrium conditions around the deterministic steady state. The subsections below provide more details about the model equations. Appendix C reports the full set of equilibrium conditions.

3.1 Patient households

Households are infinitely-lived and solve an intertemporal utility maximization problem. Each household's preferences are represented by the following intertemporal utility function:

$$U_{t} = E_{t} \sum_{s=0}^{\infty} \beta^{t+s} \left(\ln X'_{t+s} + e^{H}_{t+s} \zeta \ln h'_{t+s} - \frac{(L'_{t+s})^{\eta}}{\eta} \right),$$
(2)

where $\beta \in (0, 1)$ is the discount factor, X'_t is habit-adjusted consumption, e^H_t is a housing shock as in Iacoviello (2015), h'_t are housing holdings, L'_t is labor supply, ζ is a housing preference parameter, and η measures the elasticity of labor with respect to the real wage. In particular, X'_t is given by:

$$X_t' = C_t' - \theta C_{t-1}', \tag{3}$$

where C'_t is the level of consumption, and $\theta \in (0, 1)$ is the degree of habit formation. We assume that habit formation is external, i.e. households take time t-1 (average) consumption as given when they maximize the intertemporal utility.⁷

Households buy consumption goods, C'_t and housing, h'_t . The relative price of

⁷To make notation simpler we do not distinguish between average and individual consumption level, which are the same in the symmetric equilibrium.

housing is q_t . In addition, they invest in riskless private bonds, B'_t , and in nominal government bond holdings, B^G_t ; pay a mixture of lump-sum, τ^L_t , and distortionary taxes, τ^C_t and τ^W_t , on consumption and labor income, respectively. Each household receives: (i) the hourly wage, W'_t ; (ii) the nominal return on private bond holdings, R_t ; (iii) the nominal return on government bond holdings, R^G_t , discounted at the *ex-ante* expected haircut rate, Δ^G_t ; and (iv) government transfers, Ξ_t . Therefore, the households' budget constraint is:

$$(1 + \tau_t^C) C_t' + q_t \Delta h_t' + \frac{B_t'}{P_t} + \frac{B_t^G}{P_t} + \tau_t^L$$

$$\leq (1 - \tau_t^W) \frac{W_t'}{P_t} L_t' + \frac{R_{t-1}B_{t-1}'}{P_t} + (1 - \Delta_t^G) \frac{R_{t-1}^G B_{t-1}^G}{P_t} + \Xi_t, \qquad (4)$$

where P_t is the price level.

3.2 Impatient households

Impatient households choose consumption, C''_t , housing, h''_t , and labor, L''_t , to maximize the following inter-temporal utility function:

$$E_t \sum_{s=0}^{\infty} \left(\beta''\right)^{t+s} \left(\ln X''_{t+s} + e_t^H \zeta \ln h''_{t+s} - \frac{\left(L''_{t+s}\right)^{\eta}}{\eta} \right), \tag{5}$$

where $\beta'' < \beta$ is the discount factor, and the external habit-adjusted consumption, X''_t , is given by:

$$X_t'' = C_t'' - \theta C_{t-1}''.$$
(6)

Impatient households face two constraints in their optimization problem. First, in deciding their expenditure and work/leisure plans they need to comply with the following flow of funds equation:

$$(1 + \tau_t^C) C_t'' + q_t \Delta h_t'' + \frac{R_{t-1}B_{t-1}''}{\Pi_t} + \frac{R_{t-1}B_{g,t-1}''}{\Pi_t} \leq (1 - \tau_t^W) \frac{W_t''}{P_t} L_t'' + B_t'' + B_{g,t}'',$$
(7)

where B''_t is what they borrow from patient households, $B''_{g,t}$ denotes the amount of credit received if the government chooses to mitigate deleveraging in the private sector, $\Pi_t \equiv P_t/P_{t-1}$ is the gross inflation rate, and W''_t is their wage rate. The interest rate paid to the government is the market rate, R_{t-1} .

Second, impatient households face a limit on their obligations towards patient households arising from the fact that, if borrowers repudiate their debt obligations, lenders have the ability to repossess their assets minus a proportional transaction cost. Therefore, impatient households face a borrowing constraint, which limits what they can borrow to a fraction of the present discounted value of housing holdings:

$$B_t'' \le m'' E_t \left[\frac{q_{t+1} h_t'' \Pi_{t+1}}{R_t} \right].$$
 (8)

The interesting case is a steady state in which the return to savings is above the interest rate. In such a case, borrowing constraint (8) holds with equality and ensures that private borrowing by impatient households, B''_t , equals the present discounted value of housing holdings. As such, parameter m'' denotes the loanto-value (LTV) ratio. Moreover, $\beta'' < \beta$ ensures that impatient households will not postpone consumption and accumulate enough wealth to make the borrowing constraint not binding.

3.3 Entrepreneurs

Entrepreneurs are distributed over the unit interval $e \in (0, 1)$ and produce a differentiated goods $Y_{e,t}$ using households' labor, capital and housing as inputs. They operate under monopolistic competition, facing a Dixit-Stiglitz firm-specific demand:

$$Y_{e,t} = \left(\frac{P_{e,t}}{P_t}\right)^{-e_t^P \chi} Y_t,\tag{9}$$

where χ is the intertemporal elasticity of substitution across varieties of goods, and e_t^P is an inflation shock.

Their production function specializes as:

$$Y_{e,t} = e_t^A K_{e,t-1}^{\omega} h_{e,t-1}^{\nu} \left(L_{e,t}' \right)^{\alpha(1-\omega-\nu)} \left(L_{e,t}'' \right)^{(1-\alpha)(1-\omega-\nu)}, \tag{10}$$

where $K_{e,t}$ is capital, $h_{e,t}$ is the real estate input, and $L'_{e,t}$ and $L''_{e,t}$ are the labor inputs provided by patient and impatient households, respectively,⁸ and e_t^A is a

⁸We assume that hours of the two households enter the production function in a Cobb-Douglas fashion, as standard in this literature (see, e.g., Iacoviello, 2005; Gerali et al., 2010; Iacoviello and

technology shock. While parameters ω and ν are the elasticities of output to capital and real estate, respectively, α represents the contribution of patient households to the labor share.

Like impatient households, entrepreneurs discount the future more heavily than patient households. Hence the discount factor of the former is lower than that of the latter, $\gamma < \beta$. This means that entrepreneurs borrow as well. Entrepreneurs only care about their own consumption, $C_{e,t}$, and maximize the following inter-temporal utility function:

$$U_t = E_t \sum_{s=0}^{\infty} \gamma^{t+s} \ln \left(X_{e,t+s} \right), \qquad (11)$$

where external habit-adjusted consumption, $X_{e,t}$, is given by:

$$X_{e,t} = C_{e,t} - \theta C_{e,t-1},\tag{12}$$

subject to the entrepreneurial flow of funds:

$$\frac{P_{e,t}}{P_t}Y_{e,t} + B_{e,t} + B_{ge,t} = \left(1 + \tau_t^C\right)C_{e,t} + q_t\Delta h_{e,t} + \frac{R_{t-1}B_{e,t-1}}{\Pi_t} \\
+ \frac{R_{t-1}B_{ge,t-1}}{\Pi_t} + w_t'L_{e,t}' + w_t''L_{e,t}'' + I_{e,t} + \xi_{K,t} + \xi_{P,t}, \quad (13)$$

where $w'_t \equiv \frac{W'_t}{P_t}$; $w''_t \equiv \frac{W''_t}{P_t}$; $B_{e,t}$ represents their debt obligations towards private agents; $B_{ge,t}$ is the credit directly intermediated by the government in case of targeted intervention (analogously to the case of impatient households); $I_{e,t}$ is investment in capital goods following law of motion:

$$I_{e,t} = K_{e,t} - (1 - \delta) K_{e,t-1},$$
(14)

and $\xi_{K,t} \equiv \frac{\psi_K}{2\delta} \left(\frac{I_{e,t}}{K_{e,t-1}} - \delta \right)^2 K_{e,t-1}$ and $\xi_{P,t} \equiv \frac{\psi_P}{2} \left(\frac{P_{e,t}}{P_{e,t-1}} - 1 \right)^2 Y_t$ are quadratic costs of adjusting the capital stock and resetting the price level, respectively.

Neri, 2010; Angelini et al., 2014; Justiniano et al., 2015; Alpanda and Zubairy, 2017; Guerrieri and Iacoviello, 2017; Lambertini et al., 2017; Cantelmo and Melina, 2018, among others). This implies imperfect substitutability across the labor skills of the two groups and allows obtaining a closed-form solution for the steady state of the model. As noted by Iacoviello and Neri (2010), perfect substitutability of labor inputs causes a more complex interaction between the borrowing constraint of impatient households and the labor supply decision, hence our choice to adopt a Cobb-Douglas specification.

Also entrepreneurs face a limit on their obligations towards patient households:⁹

$$B_{e,t} \le mE_t \left[\frac{q_{t+1}h_{e,t}\Pi_{t+1}}{R_t}\right].$$
(15)

3.4 Government

The government finances its expenditures, G_t , by levying taxes, T_t , and by issuing bonds, B_t^G . It promises to repay one-period bonds the next period and the gross nominal interest rate applied is R_t^G . However, in order to introduce a sovereign risk premium, we assume that government bond contracts are not enforceable, and use to that end the same modeling strategy as in Cantore et al. (2018). Each period, a stochastic fiscal limit expressed in terms of government debt-to-GDP ratio and denoted by Γ_t^* , is drawn from a distribution, the cumulative density function (CDF) of which is represented by a logistic function, p_t^* , with parameters η_1 and η_2 :

$$p_t^* = P\left(\Gamma_t^* \le \Gamma_t\right) = \frac{\exp\left(\eta_1 + \eta_2 \Gamma_t\right)}{1 + \exp\left(\eta_1 + \eta_2 \Gamma_t\right)},\tag{16}$$

where $\Gamma_t \equiv B_t^G/Y_t$.¹⁰ If government-debt-to-GDP exceeds the fiscal limit, i.e. $\Gamma_t \geq \Gamma_t^*$, then the government defaults. Hence p_t^* represents the probability of default. This occurs in the form of an haircut $\Delta_t^G \in [0, 1]$ applied as a proportion to the outstanding stock of government debt. Agents consider the *ex-ante* expected haircut rate,

$$\Delta_t^G = \begin{cases} 0 & \text{with probability } 1 - p_t^* \\ \bar{\Delta}^G & \text{with probability } p_t^* \end{cases},$$
(17)

where $\bar{\Delta}^G \in (0, 1]$ is the haircut rate applied in the case of default.¹¹ In other words:

$$\Delta_t^G = p_t^* \bar{\Delta}^G. \tag{18}$$

This mechanism is akin to that of Ghosh et al. (2013) who determine a "debt limit" beyond which fiscal solvency is in doubt. Following what they call "primary

 $^{^{9}}$ Following Iacoviello (2005) we assume that, just like households, firms only use housing as collateral. For a model in which physical capital is also a collateralized asset see Liu et al. (2013).

¹⁰Also Bi and Traum (2014) assume the same functional form for the CDF of the fiscal limit.

¹¹While the model captures a convex increase in the risk premium as the level of government debt approaches the fiscal limit, it is solved within a region of fiscal solvency, with possibly a very small fiscal space.

balance rule", governments increase primary (i.e. non-interest) fiscal surpluses in response to rising debt service, so that the public debt-to-GDP ratio will converge to its long-run value. This fiscal reaction function - analogous to our equations (23) and (24) below – is consistent with the empirical findings of Mendoza and Ostry (2008) for a sample of industrial and emerging market economies.¹² In response to shocks, the debt ratio will be stabilized if the primary balance increases to offset the higher interest rate bill. However, at sufficiently high levels of debt, the occurrence of large shocks may require a very sizable increase in the primary balance. Even assuming an interest rate inelastic to government indebtedness, there will be a debt level beyond which dynamics turn explosive and the government will default. In fact, default will occur at a level of debt below this threshold because, as the probability of default rises, financial market participants will require higher and higher risk premia, making it more unlikely that the primary surplus will be enough to meet debt obligations, and in turn increasing the default probability even further. Azariadis (2016), among others, indeed shows that expectations regarding future debt prices affect public debt sustainability.

Furthermore, Ghosh et al. (2013) empirically find that other factors determining fiscal limits are economic growth, inflation, trade openness and the country's structural characteristics such as political stability and institutionalized fiscal rules. It should also be noted that, according to conventional wisdom, political economy considerations may hinder fiscal adjustments, although Alesina et al. (2012) find no evidence of systematic turnover of governments implementing large reductions of budget deficits.

In sum, both financial market reactions and policymakers' decisions are surrounded by uncertainty, which makes fiscal limits far from being deterministic. Therefore, assuming stochastic fiscal limits that depend on the debt ratio seems a reasonable and tractable way to capture these complex phenomena.

In some experiments, we give the government the option to conduct "targeted interventions", namely to intermediate funds towards financially constrained agents, using a mechanism similar to that proposed by Gertler and Karadi (2011) for unconventional monetary policy. In the case of a targeted intervention, the government issues additional bonds $B_t^{int} \equiv B_{g,t}'' + B_{g,t}$, that pay the gross nominal interest rate

¹²The response of the primary balance to debt is estimated to be stronger in emerging economies than in advanced countries, likely because the riskier financial and fiscal conditions of the former make a stronger response necessary to maintain fiscal solvency.

 R_t^G , and lends the raised funds to the private sector at the market rate R_t . This operation is assumed to imply some inefficiency costs equal to κ per unit supplied capturing the fact that public intermediation is likely to be less efficient than private intermediation. The inefficiency cost affecting the government budget constraint is then $\Upsilon_t \equiv \kappa B_t^{int}$, which is a dead weight loss.

Simple rules define how the government intervention takes place, and link government intervention to deleveraging, to an extent controlled by parameter ϵ :

$$b_{g,t}'' = -\epsilon b_t'',\tag{19}$$

$$b_{g,t} = -\epsilon b_t,\tag{20}$$

where lower-case letters indicate deviations of debt variables from their respective steady state, relative to steady-state output, $x_t \equiv \frac{X_t - X}{Y}$. We assume that, at the steady state, no government intervention occurs $(B''_g = B_g = 0)$. If $\epsilon = 0$, the model collapses to the standard case in which funds are intermediated only by the private sector.

One key point of departure from the mechanism of Gertler and Karadi (2011) relates to the assumption that in our model the government is subject to fiscal limits and that these generate a sovereign risk premium. This assumption introduces an additional cost, given by the spread $(R_t^G - R_t)$ times the units of funds intermediated B_t^{int} , in the government flow of funds:

$$B_t^G = \left(1 - \Delta_t^G\right) \frac{R_{t-1}^G B_{t-1}^G}{\Pi_t} + G_t + \frac{\left(R_{t-1}^G - R_{t-1}\right) B_{t-1}^{int}}{\Pi_t} + \Upsilon_t - T_t + \Xi_t.$$
(21)

Note that the government borrows from patients agents and lends to impatient agents within the same period. Thus, the total cost of targeted intervention is given by the sum of the inefficiency cost and the increment in debt servicing costs associated with the additional borrowing needed to extend credit to the private sector. Also note that each period transfers are set in a way that sovereign default does not alter the actual debt level, $\Xi_t \equiv \Delta_t^G \frac{R_{t-1}^G B_{t-1}^G}{\Pi_t}$.¹³

 $^{^{13}}$ This approach is also followed by Corsetti et al. (2013). The absence of such transfers would imply lower risk premia prior to default, as the lower post-default debt stock would already be taken into account.

Total government revenue T_t is given by:

$$T_t = \tau_t^C \left(C'_t + C''_t + C_t \right) + \tau_t^W \left(w'_t L'_t + w''_t L''_t \right) + \tau_t^L.$$
(22)

In order to reduce the number of tax instruments to one, we impose that τ_t^C , τ_t^W and τ_t^L deviate from their respective steady state by the same proportion (i.e. $\tau_t^C = \tau_t \bar{\tau}^C$, $\tau_t^W = \tau_t \bar{\tau}^W$, $\tau_t^L = \tau_t \bar{\tau}^L$), and that the proportional uniform tax change, τ_t , becomes one of our fiscal policy instruments. As common in the literature, the steady-state value of the lump-sum tax is treated as a residual to calibrate the government debt at a desired steady-state level.

We allow the tax and government spending instruments to be adjusted according to the following feedback rules:

$$\log\left(\frac{\tau_t}{\tau}\right) = \rho \log\left(\frac{\tau_{t-1}}{\tau}\right) + (1-\rho) \left[e^{\phi \frac{B^G}{Y}} \rho_B \log\left(\frac{B^G_{t-1}}{B^G}\right)\right], \quad (23)$$

$$\log\left(\frac{G_t}{G}\right) = \rho \log\left(\frac{G_{t-1}}{G}\right) - (1-\rho) \left[e^{\phi \frac{B^G}{Y}} \rho_B \log\left(\frac{B_{t-1}^G}{B^G}\right)\right], \quad (24)$$

where ρ implies persistence in the fiscal policy instruments; ρ_B is the responsiveness of the instruments to the percent deviation of government debt from its steady state; and $e^{\phi \frac{B^G}{Y}}$ is an exponential factor tightening the fiscal policy stance for increasing steady-state levels of the government debt-to-GDP ratio.

Some remarks on the above fiscal rules are in order. First, these rules imply that fiscal instruments are set to keep public debt under control, as generally observed in advanced economies (see Mendoza and Ostry, 2008, among others). Second, the introduction of persistence captures policy inertia, which is justified on empirical grounds. For instance, Forni et al. (2009) impose very similar fiscal rules and find evidence of substantial policy inertia in a model estimated using euro area data. Third, one feature of our model is that higher levels of public debt, via an increase in the sovereign default probability and the associated risk premium, imply higher debt servicing costs and make debt dynamics more prone to instability. It follows that higher public indebtedness requires a stronger responsiveness of the fiscal instruments to public debt. This mechanism is line with the empirical findings of Bohn (1998, 2005) who identifies a stronger response of the primary balance to debt at high debt ratios. We capture this channel via the exponential term $e^{\phi \frac{B^G}{Y}}$, which is useful also for conducting our numerical exercises because, once parameters ρ_B and ϕ have been set, it automatically implies a tighter fiscal stance at higher steady-state levels of government debt.

Although in practice the government may exhibit different degrees of inertia and elasticities for different instruments, assuming the same parameters for all fiscal instruments greatly simplifies the exercises presented in the following sections without loss of generality.¹⁴

3.5 Central bank

Monetary policy is set according to a Taylor-type interest-rate rule,

$$\log\left(\frac{R_t}{R}\right) = \rho_\pi \log\left(\frac{\Pi_t}{\Pi}\right) + \rho_y \log\left(\frac{Y_t}{Y}\right),\tag{25}$$

where ρ_{π} and ρ_y are the monetary policy parameters that dictate the responsiveness of the interest rate to deviations of inflation and output from their respective steadystate values.¹⁵

3.6 Equilibrium

Equilibrium in the goods market, the loans market, and the housing market implies that $Y_t = C_t + C'_t + C''_t + I_t + G_t + \Upsilon_t + \xi_{P,t} + \xi_{K,t}$; $B_t + B'_t + B''_t = 0$, where $B_t = \int_0^1 B_{e,t} de$; and h + h' + h'' = 1. This last equilibrium condition in turn implies that housing is in fixed supply, which we normalize to one. The model is completed by autoregressive processes for the shocks, $\log\left(\frac{e_t^{\varkappa}}{e^{\varkappa}}\right) = \rho_{\varkappa} \log\left(\frac{e_{t-1}^{\varkappa}}{e^{\varkappa}}\right) + \epsilon_t^{\varkappa}$, where $\varkappa = \{A, H, P\}$, ρ_{\varkappa} are autoregressive parameters and ϵ_t^{\varkappa} are mean zero, i.i.d. random shocks with standard deviation σ^{\varkappa} .

4 Model calibration and validation

Table 3 reports the parameter values used to simulate the model. To the extent possible, we set our baseline calibration in line with previous empirical estimates or stylized facts on the Euro Area. Where this is not possible, for example agents' relative discount factors, we take parameter values from studies on the United States.

 $^{^{14}}$ For papers assessing the effects of fiscal policy in a multi-country dynamic general equilibrium model of the euro area see Gomes et al. (2012, 2016), among others.

¹⁵Modeling the zero lower bound is beyond the scope of the paper.

Parameter		Value
Patient households' discount factor	β	0.99
Impatient households' discount factor	β''	0.95
Entrepreneurs' discount factor	γ	0.98
Labor supply elasticity	η	1.01
Habits in consumption	heta	0.592
Capital depreciation rate	δ	0.03
Capital share	ω	0.30
Patient households' wage share	α	0.64
Capital adjustment costs	ψ_K	2.00
Elasticity of substitution in goods	χ	6.00
Price stickiness	ψ_P	41.667
Inflation - Taylor rule	$ ho_{\pi}$	1.50
Output -Taylor rule	$ ho_y$	0.10
SS stock of residential housing over annual output	$\bar{q}\left(\bar{h}'+\bar{h}''\right)/\left(4\bar{Y} ight)$	1.34
SS commercial real estate over annual output	$\bar{q}\bar{h}/\left(4\bar{Y} ight)$	0.65
SS share of government spending in GDP	\bar{G}/\bar{Y}	0.23
SS consumption tax rate	$ar{ au}^{\dot{C}}$	0.20
SS labor income tax rate	$ar{ au}^W$	0.45
Persistence of fiscal instruments	ρ	0.90
Fiscal responsiveness to government debt	$ ho_B$	0.01
Responsiveness of the fiscal stance to government debt	ϕ	1.35
Scaling factor in default probability	η_1	-8.5527
Slope parameter in default probability	η_2	1.8261
Degree of targeted intervention	ϵ	0
Inefficiency costs	κ	0
SS impatient households loan-to-value ratio	m''	0.80
SS entrepreneurs loan-to-value ratio	m	0.375
SS public debt-to-GDP ratio	$\bar{\Gamma}/4$	0.71
Persistence of housing shock	ρ_H	0.9887
Persistence of inflation shock	ρ_P	0.8487
Persistence of technology shock	ρ_A	0.0335
Standard deviation of housing shock	σ^H	0.0105
Standard deviation of inflation shock	σ^P	0.0026
Standard deviation of technology shock	σ^A	0.0236

Table 3: Baseline Parameter Values

Shocks are calibrated to match key moments in Euro Area data. The time period in our model corresponds to one quarter in the data.

The following parameter values are rather common in the literature (see Iacoviello, 2005, among others): agents' discount factors, $\beta = 0.99$, $\beta'' = 0.95$, and $\gamma = 0.98$; capital share, $\omega = 0.30$; patient households' wage share, $\alpha = 0.64$; and capital adjustment costs, $\psi_K = 2$. The value of the Frisch elasticity of labor supply, $1/(\eta - 1)$, is a source of controversy. While most microeconomic studies suggest an estimate of the elasticity ranging from 0 to 0.5, the estimates used by the business cycle literature are much higher (for an early survey of the literature see Card, 1991). More recently Keane and Rogerson (2012), however, concluded that estimates of small labor supply elasticities based on micro data are fully consistent with large aggregate labor supply elasticities. As regards capital depreciation, the business cycle literature generally agrees on an annual rate around 10 percent. In our baseline calibration, we follow Iacoviello (2005) who assumes a very elastic labor supply $(1/(\eta - 1) = 100 \Leftrightarrow \eta = 1.01)$ and an annual capital depreciation rate of 12 percent ($\delta = 0.03$), as our model builds on his framework. However, in Appendix D we show that results are robust to different choices of both the Frisch elasticity and the capital depreciation rate.

The value of habit persistence, $\theta = 0.592$, is taken from the estimated Euro-Area model of Smets and Wouters (2003). For the interest rate response to inflation we choose a value that satisfies the Taylor principle, $\rho_{\pi} = 1.5$ (e.g. Taylor, 1993). In line with the empirical literature on the euro area (see, e.g., the influential paper by Smets and Wouters, 2003; Adolfson et al., 2007; Forni et al., 2009; Villa, 2016) we set the interest rate response to output, ρ_y , to 0.1. In Appendix D we check if results are robust to a higher value of ρ_y . For the steady-state values of the share of government spending in GDP, $\bar{G}/\bar{Y} = 0.23$, and the two distortionary tax rates, $\bar{\tau}^C = 0.20$ and $\bar{\tau}^W = 0.45$, as well as the degree of price stickiness, $\psi_P = 41.667$, we rely on the values used by Christiano et al. (2010) for the Euro Area.¹⁶ Then, in line with the data, we make fiscal instruments persistent ($\rho = 0.90$). We set the degree of fiscal stance, $\rho_B = 0.01$, and its responsiveness to government debt, $\phi = 1.35$, to approximately the minimal value needed to stabilize public debt in the range of government debt-to-GDP ratios explored. The elasticity of substitution across different varieties, χ , is equal to 6 in order to target a steady state gross mark-up equal to 1.20 as common in the literature.

The steady-state stock of residential housing over annual output, $\bar{q}(\bar{h}' + \bar{h}'')$ / $(4\bar{Y}) = 1.34$, is taken from the the OECD database on balance sheet for nonfinancial assets on households dwellings in France and Germany between 2000 and

¹⁶The value of ψ_P is chosen to match the same slope of the linearized New-Keynesian Phillips curve of Christiano et al. (2010) where prices are set as in Calvo (1983).

2013.¹⁷ Such a value is matched through an appropriate choice of ζ . The steadystate commercial real estate over annual output, $\bar{q}\bar{h}/(4\bar{Y}) = 0.65$, is taken from the OECD database on balance sheet for non-financial assets on dwellings of nonfinancial corporations in France and Germany between 2000 and 2013. Such a value is matched through an appropriate choice of ν . In the baseline case, the households' LTV ratio, m, is equal to 0.80, the typical LTV ratio for a new mortgage in the majority of the Euro Area countries in 2007 (ECB, 2009). The entrepreneurial LTV, m = 0.375, is taken from data on corporate indebtedness in the Euro Area (ECB, 2012). Last, the debt-to-GDP ratio $\bar{\Gamma}/4 = 0.71$ corresponds to the average debt-to-GDP ratio for countries in our panel dataset. Given that the parameters related to government and private indebtedness are crucial for the results, we explore sensitivity to a wide range of values in Section 5.

Moreover, we assume no targeted intervention and zero inefficiency costs, i.e. $\epsilon = \kappa = 0$. We nonetheless show how alternative values of these two parameters affect the results in Subsection 5.3.

To calibrate the CDF of the fiscal limit, we fix two points on the function in a way consistent with empirical evidence. Given two points (Γ_1, p_1^*) and (Γ_2, p_2^*) , with $\Gamma_2 > \Gamma_1$, parameters η_1 and η_2 are uniquely determined by

$$\eta_2 = \frac{1}{\Gamma_1 - \Gamma_2} \log \left(\frac{p_1^*}{p_2^*} \frac{1 - p_2^*}{1 - p_1^*} \right), \tag{26}$$

$$\eta_1 = \log\left(\frac{p_1^*}{1 - p_1^*}\right) - \eta_2 \Gamma_1.$$
(27)

Let us assume that when the ratio of government debt to annual GDP is Γ_2 , the probability of exceeding the fiscal limit is almost unity, i.e. $p_2^* = 0.99$. We set the fiscal limit at $\Gamma_2 = 4 \times 1.8$, broadly in line with Greece's recent experience during the sovereign debt crisis. Let us fix $\Gamma_1 = 4 \times 0.6$, the average general government consolidated gross debt in the United States over the period 1980-2007. Before the financial crisis, the U.S. sovereign risk premium was very small – around 15 annual basis points (ABP) for sovereign default swap spreads (see e.g. Austin and Miller, 2011). Hence we assume that for $\Gamma_1 = 4 \times 0.6$, $ABP_1 = 15$. At the onset of the Greek sovereign debt crisis, the sovereign risk premium skyrocketed to an order of

¹⁷The steady-state stock of residential housing over annual output has a similar value when considering the average of euro area countries.

magnitude of around 1,000 annual basis points, hence we fix $ABP_2 = 1,000$. The haircut rate, $\bar{\Delta}$, consistent with ABP_2 and p_2^* is obtained as $\bar{\Delta} = \left[1 - \frac{1}{\frac{ABP_2}{40000} + 1}\right] / p_2^* \cdot {}^{18}$ At this point, we can recover the probability of default when $\Gamma = \Gamma_1$,

$$p_1^* = \frac{1 - \frac{1}{\frac{ABP_1}{40000} + 1}}{\bar{\Delta}},$$

which is $p_1^* = 0.0152$, and the parameters η_1 and η_2 of the fiscal limit CDF can be recovered by using equations (26) and (27), i.e. $\eta_1 = -8.5527$ and $\eta_2 = 1.8261$. This parametrization implies that the probability of default remains moderate (below 20%) until the government debt-to-annual-GDP is below 100%, and then it increases at an expedited rate. This captures the fact that the pricing of risk of sovereign bonds for a country with weak fiscal fundamentals, can deteriorate rapidly and nonlinearly as public debt accumulates. Analogous modeling of the risk premium can be found in Bi (2012), and is consistent with the empirical relationship observed in a a large set of advanced countries between government gross debt and sovereign spreads as reported by Corsetti et al. (2013).

Last, we set the standard deviations and the persistence of the shocks via moment-matching of the empirical standard deviations and the persistence of real output, inflation and the real house price.

Given the difficulty in matching all moments precisely, we construct a quadratic loss function $L = \sum_{j=1}^{6} (x_j^m - x_j^d)^2$, where x_j^m is the *j*-th moment in the model and x_j^d is its analogue in the data, and we numerically search for those parameters that minimize *L*. This procedure leads to persistent housing and inflation shocks, $\rho_H =$ 0.9887 and $\rho_P = 0.8487$; while, as in Iacoviello (2005), the estimated technology shock exhibits a small persistence $\rho_A = 0.0347$, as the model produces significant endogenous persistence. The standard deviations of the shocks are around 1% and 2% in magnitude.

Table 4 shows the volatilities, persistences and correlations of variables in the data and in the model that we directly target, as well as two other important

$$\frac{R^G}{R} = \frac{1}{(1 - \Delta^G)} = 1 + \frac{ABP}{40000},$$

 $^{^{18}}$ To see this, note that equations (C.3) and (C.4) imply the following steady-state sovereign risk premium:

using which Δ^g can be written as a function of a chosen premium expressed in annual basis points, $\Delta^g = 1 - \frac{1}{1 + \frac{ABP}{1 + \frac{ABP}{40000}}}$. Finally, from equation (18) $\bar{\Delta}^G = \Delta^G / p^*$.

Moment	Data	Model
Standard deviations		
Real output	0.0138	0.0088
Inflation	0.0061	0.0047
Real house prices	0.0158	0.0193
Autocorrelations		
Real output	0.8779	0.9435
Inflation	0.2386	0.2616
Real house prices	0.8614	0.8658
Cross-correlations with output		
Investment	0.8221	0.9728
Private consumption	0.9218	0.9955

Table 4: Moments of Key Macroeconomic Variables

moments.¹⁹ As the table indicates, with this calibration the model replicates reasonably well the moments in the data, and also manages to approximate closely to the cross-correlations of investment and private consumption with output.

To check the extent to which our model is able to replicate observed historical patterns, we simulate time series of output, private and public debt-to-GDP ratios. We then estimate on these simulated data the same regressions run on actual data (Table 1).²⁰ To facilitate the comparison between the two sets of estimates, we simulate the model under calibrations of public debt-to-GDP using the same values for average, "low" and "high" ratios reported in Section 2. Therefore, we set $\overline{\Gamma}/4 = 0.71$, $\overline{\Gamma}/4 = 0.54$ and $\overline{\Gamma}/4 = 0.92$, to capture average, "low" and "high" public debt cases, respectively.

Table 5 shows that the model-generated data displays correlations with the same signs as those associated with actual Euro Area data. Simulated times series also capture the nonlinearities identified in the empirical relationship between public debt and output for "low" versus "high"-public-debt countries.²¹

¹⁹Data on euro area countries are taken from the Statistical Data Warehouse of the ECB and the International Financial Statistics database of the IMF. They refer to the period 1999Q1-2015Q1 (or shorter where observations are not available). Time series of GDP components and real house prices are detrended using the HP filter.

 $^{^{20}}$ To take advantage of the illustrative simulation exercise, we consider a large sample of simulated observations equal to 5,000 quarters for each regression.

²¹The estimated coefficients are always significant at a 1 percent level due to the large size of the simulated sample. Hence significance stars and standard errors are not reported.

	Dependent variable: \hat{y}_{it+12}			
	(1)	(2)	(3)	(4)
$\left(\frac{P\hat{R}D}{Y}\right)_t$	-0.180	-0.191	-0.216	-0.070
$\left(\frac{P\hat{U}D}{Y}\right)_t$		0.040	0.049	-0.022
"Low" public debt	\checkmark	\checkmark	\checkmark	
"High" public debt	\checkmark	\checkmark		\checkmark
R^2	0.029	0.031	0.036	0.025
Observations	5000	5000	5000	5000

Table 5: Cyclical Fluctuations of Private and Public Debt and Subsequent Cyclical Fluctuations of Real GDP in Simulated Data

Notes: Estimates are obtained via regressions of simulated series of real GDP in t+3 on simulated series of private and public debt to GDP ratios in t.

5 A deleveraging episode

The results presented in the previous sections are obtained assuming that all shocks (both in actual and simulated data) are at play simultaneously. In this section we focus on the effects of a single shock, namely a negative house price disturbance that triggers a deleveraging episode. Specifically, we explore the macroeconomic consequences of deleveraging (Subsection 5.1), the role that private and public debt have in affecting economic activity during a deleveraging phase (Subsection 5.2), and the mitigating effects of government targeted intervention, as well as how these depend also on the size of fiscal buffers (Subsection 5.3).

5.1 The macroeconomic consequences of deleveraging

We trigger a downward phase of a leverage cycle via a temporary negative shock to households' housing preferences. The shock generates a decline in house prices, which in turn depresses the value of the housing collateral. In the experiments discussed throughout, the shock is set to induce a fall in house prices equivalent to one percent. Although the shock originates in the households' side of the model, the induced decrease in house prices affects also firms' borrowing constraints. As a result, both households' and corporate debt react to the shock.

Figure 1 shows that the protracted decline in house prices, and the consequent fall in the value of constrained agents' collateral make borrowing constraints tighter. This forces private agents to deleverage by cutting consumption and investment. The



Figure 1: Impulse Responses to a Negative One-Per-Cent House Price Shock

Notes: X-axes in quarters; Y-axes are in percent deviations from steady state, except for private and public debt to GDP ratios where deviations are absolute.

fall in private demand implies a protracted output contraction and a deflation. The size of the output response matches well the observed relationship between changes in house prices and the output gap in the Euro Area, e.g. during the sovereign debt crisis of 2010-2013.²² The fall in households' debt is also more persistent than the decline in corporate debt, in line with observed developments based on BIS data of credit to households and non-financial corporations in the Euro Area in the aftermath of the past recession (this stylized fact is documented by Smets and Villa, 2016, also for the U.S.). The worsened economic outlook spills over to public finances: the fall in output induces a reduction of government revenues and the public debt-to-GDP ratio rises unambiguously. This mechanism is enhanced by

²²The deviation of the Eurostat House Price Index reached a maximum of 4% below its trend and the euro area output gap reached a trough of about 1% (2013q1), a level close to what the model would suggest, $\approx (0.17 \times 4) \% = 0.7\%$.

debt deflation, and by the fact that higher public indebtedness boosts the sovereign risk premium, propping up government's financing costs.²³

5.2 The role of private and public debt during deleveraging

To understand the extent to which public and private indebtedness matters, in Figure 2 we simulate the model under varying assumptions about the steady-state level of the private and public debt/GDP ratios. Namely, we vary private debt at the steady state by modifying the loan-to-value (LTV) ratio parameters. To shift the steady-state value of public debt, we change the long-run target of the public debt-to-GDP ratio.²⁴ We plot the three-year cumulative responses of output, private and public debt-to-GDP ratios to an identical negative house price shock.

There are three main results. First, the economic contraction is increasingly worse the higher private debt is relative to GDP. In other words, countries in which households and firms have borrowed more to finance expenditure will experience a larger drop in demand when credit constraints become tighter. This result is in line with the standard behavior of the financial accelerator effect embedded in these models featuring collateral constraints (e.g. Iacoviello and Neri, 2010; Liu et al., 2013; Christensen et al., 2016, among many others).

Second, due to the decline in output, government fiscal revenues in countries with higher initial private debt decline by more and more rapidly when private credit dries up. Together with the greater severity of the drop in GDP, this leads

²³Note that Figure 1 shows impulse response functions to the house price shock for 20 quarters. Table 5, instead, shows the correlations between cyclical fluctuations of private and public debt and cyclical fluctuations of real GDP 12 quarters-ahead, using a large sample of 5,000 observations. In response to the house price shock the correlation between cyclical fluctuations of public debt and cyclical fluctuations of real GDP 12 quarters-ahead is negative even at an horizon longer than the one reported in the figure. The positive (and low) correlation between these two variables shown in Table 5 is driven mainly by the technology shock, which plays an important role in accounting for fluctuations in public debt.

²⁴Specifically, to produce the charts in the left column of the figure, we keep the steady-state level of public debt/GDP at the baseline value of $\Gamma = 4 \times 0.71$. This is equivalent to the average level of public debt in our Euro Area dataset, i.e. 71 percent of annual GDP. Then, we set the two LTV ratios (for impatient households and firms) to the same value, let them vary over the same parameter range ($m = m'' \in [0.375, 0.80]$), and we plot the corresponding three-year cumulative responses of output, public debt/GDP and private debt/GDP. For ease of comparison with public debt, on the x-axis we report the resulting private debt/GDP ratio. Instead, to produce the charts in the right column of the figure, we keep the steady state of private debt at the baseline level (m = 0.375 and m'' = 0.80), let the steady-state level of public debt/GDP vary in the interval $\overline{\Gamma} \in [4 \times 0.5, 4 \times 1.2]$, so that public debt ranges between 50 and 120 percent of annual GDP, and plot the same variables.

Figure 2: Three-Year Cumulative Responses to a Negative One-Per-Cent House Price Shock for Different Loan-to-Value (LTV) Ratios and Different Steady-State (SS) Public Debt/GDP Ratios



Notes: In the left column the LTV ratios, m and m'', vary between 0.375 and 0.80; for ease of comparison with public debt, on the x-axis we report the resulting private debt/GDP ratio. In the right column, the steady-state government debt-to-GDP ratio, Γ , varies between 4×0.5 and 4×1.2 ; Y-axes for output are in percent deviations from steady state, while for private and public debt to GDP ratios they are in absolute deviations.

to a more pronounced increase in public debt as a fraction of GDP.

The third main result is that the higher public debt relative to GDP before the financial shock, the worse the recession, but this relationship becomes material only when public debt is above a certain threshold (about 90 percent of GDP). The behavior of these impulse responses highlights the non-linear relationship between the average level of public debt and the severity of the contraction. This can be explained by the fact that any change in public debt implies a different elasticity of the return on government bond holding, R^G , to public debt given the assumed CDF of the fiscal limit.²⁵

The relationship between steady-state public debt levels in terms of GDP and its cumulative response is positive but also non-linear: once again, for levels of the steady-state public debt-to-GDP ratio below 0.90 the cumulative response of public indebtedness is marginally affected. The effects become stronger at higher levels of steady-state public debt that, in line with the calibration of the CDF of the fiscal limit, imply a boost in sovereign risk premia and the government's financing costs.

Interestingly, differences in levels of private debt exert proportionally stronger impacts on output than differences in public debt. For example, an increase in the steady-state level of private debt-to-GDP ratio by 10 percentage points leads to a deepening three-year GDP contraction of 11 percent on average. The worsening of the recession is approximately proportional to the increase in the private debtto-GDP ratio. Instead, nonlinearities are more evident for the case of public debt. When the level of steady-state public debt is lower than 90 percent of GDP, an increase of the ratio by 10 percentage points leads to a deepening of the three-year GDP contraction of less than 1 percent. The worsening in the GDP contraction reaches a maximum value of 6 percent when steady-state public debt increases from 100 to 110 percent of GDP. The rationale for this result is that the more impatient agents have resorted to borrowing to finance their expenditure, the larger the required spending cut to satisfy their borrowing constraints when the shock hits. In contrast, if the government is sufficiently far from its fiscal limit, it can borrow more and partially offset the financial shock, even in the face of higher financing costs.²⁶

²⁵Linearizing the relevant equilibrium conditions, it is possible to derive the elasticity of the return on government bond holding to public debt, Θ_t . Note that $\hat{R}_{t-1}^G = \frac{\Delta^G}{(1-\Delta^G)} \frac{\eta_2}{1+A(\Gamma)} \Gamma \hat{\Gamma}_t + \hat{R}_{t-1}$, hence the elasticity $\Theta_t = \frac{\Delta^G}{(1-\Delta^G)} \frac{\eta_2}{1+A(\Gamma)} \Gamma$, where $A(\Gamma) = exp(\eta_1 + \eta_2\Gamma)$. This elasticity inherits the property of the first derivative of the CDF, which is a convex function of Γ .

 $^{^{26}}$ This case is reminiscent of the point made by Ostry et al. (2015), whereby if public debt is sufficiently below that implied by the fiscal limit, the government is still better off increasing its debt further to absorb a negative shock.

5.3 The effects of targeted intervention and the merits of fiscal buffers

In our model, the government can lend money to the private sector (i.e. impatient households and entrepreneurs) at times when swings in the value of their collateral impose sharp and abrupt cuts to the credit they have access to. We refer to this policy as "targeted intervention". We interpret the intervention in a general sense, to capture a set of real-world policy measures adopted by various governments in the Euro Area and elsewhere during the global financial crisis to alleviate households' and firms' borrowing constraints. These initiatives included, among others, government credit to facilitate mortgage payments by agents in distress (for example in Spain), unsecured government lending to households to finance home renovation projects, credit support to firms (for example in Germany and France) at a time when other forms of financial intermediation had dried up.²⁷

For the government there is an obvious merit in relaxing the private sector's borrowing constraints at times of stress: similarly to unconventional monetary easing, by lending directly to the larger economy, the government can effectively remove a financial friction that would otherwise impede credit-constrained agents to smooth spending during a deleveraging phase. In so doing, the government *de facto* supports economic activity while, at the same time, it mitigates the drop in government revenues. We show below that, if the policy is appropriately set, it has dual beneficial macroeconomic consequences because it averts a deterioration of the recession as well as a deterioration of public finances, including public debt.

Our set up is well suited to clarify how these mechanisms work because it embeds explicitly the links between private and public debt, and output. To be worthwhile, the output/fiscal revenue gains from targeted interventions must thus be large

²⁷The fiscal stimulus implemented by Euro Area governments in response to the financial crisis is known as the European Economic Recovery Plan (EERP). These fiscal measures have been implemented in addition to the stimulus provided via automatic fiscal stabilizers. One of the key principles of the EERP is that the budgetary stimulus should be *targeted* towards the source of the economic challenge (increasing unemployment, credit constrained firms/households, etc. and supporting structural reforms) (Commission, 2008, page 8), within the Stability and Growth Pact. Among the measures listed in the EERP, "guarantees and loan subsidies to compensate for the unusually high current risk premium can be particularly effective in an environment where credit is generally constrained". The enhanced access to financing for business is implemented also via the intervention of the European Investment Bank, which has put together a package of 30 billion euro for loans to small and medium enterprises, an increase by 10 billion euro over its usual lending in this sector.



Figure 3: Impulse Responses to a Negative One-Per-Cent House Price Shock: Effects of Targeted Intervention

Notes: Targeted intervention is obtained by setting $\epsilon = 0.1$ and $\kappa = 0.1$; X-axes in quarters; Y-axes are in percent deviations from steady state, except for private and public debt to GDP ratios where deviations are absolute.

enough to outweigh the adverse impact on output of subsequent necessary fiscal consolidations aimed at keeping public debt sustainable. This trade-off depends on the amount of fiscal space (namely the distance between the initial stock of outstanding government debt and the fiscal limit) available when the shock hits. The smaller the fiscal space (i.e. the larger public debt) before the financial crisis, the larger the debt servicing costs, the narrower the room for maneuver for targeted intervention, the tighter the fiscal policy stance needs to be.²⁸

²⁸By contrast, the model abstracts from the potential implications of moral hazard, that is the possibility that impatient agents may borrow excessively, factoring in the likelihood of a future government targeted intervention in their favor. Below we partially compensate for this by introducing inefficiency costs to government targeted intervention. While this does not address the simplification of ruling out moral hazard behavior, they do approximate the economic losses that may potentially arise from such behavior.

In Figure 3 we examine the role of targeted intervention in the simulated deleveraging episode. To activate this policy channel in the model we set the relevant parameters ϵ and κ equal to 0.1. This parametrization captures a moderate degree of targeted intervention and inefficiency costs in the range considered by Gertler and Karadi (2011). While in this exercise the degree of targeted intervention is set arbitrarily, in the remainder of this subsection we compute the level of intervention that maximizes gains and explore the interactive effects of ϵ and κ on the effectiveness of the policy.

The illustrative simulation shows that government lending is effective at containing the contractionary consequences of private sector deleveraging. This, in turn, helps contain the decline in government revenues and the extent of the deflation. However, the intervention raises public spending temporarily, because of the increase in the risk premium, and thus in debt servicing costs, and the cost of inefficiencies arising from public intermediation. Reflecting these additional costs, public debt is not only higher but also more persistent compared to the baseline scenario.

By choosing the level of targeted intervention (ϵ) appropriately, however, the government can simultaneously support economic activity and reduce the debt-to-GDP ratio. To see how, let us suppose that the private sector is highly indebted, which we capture by setting the LTV ratios at levels located in the highest range of the historical Euro Area statistical distribution (ECB, 2012) – m'' = 0.99 and m = 0.44. Let us also suppose that the government has fiscal space, so that the targeted intervention does not need to be compensated off through an aggressive exacerbation of the fiscal stance (to this end we set the steady-state level of public debt/GDP at the average Euro Area value of 71 percent, i.e. $\bar{\Gamma} = 4 \times 0.71$). To check whether and to what extent it is desirable for the government to intervene, we compute the gains from targeted intervention, i.e. the differences between threeyear cumulative impulse responses to a negative one-per-cent house price shock for different degrees of targeted intervention ($\epsilon \in [0,1]$) and their analogues in a no-policy-action scenario ($\epsilon = 0$). We conduct the same exercise for three levels of inefficiency costs ($\kappa = \{0, 0.1, 0.5\}$) and report the results in Figure 4. This parameter range for κ is very wide: $\kappa = 0.1$ is already in the high range of values that Gertler and Karadi (2011) consider plausible; $\kappa = 0.5$ captures the case of extreme inefficiencies in public intermediation of funds.²⁹

²⁹Throughout this subsection, in analogy to Clerc et al. (2015), we consider a measure of output directly comparable with the data, which is net of adjustment costs and dead weight losses, i.e.



Figure 4: Three-Year Cumulative Gains from Targeted Intervention Relative to a No-Policy-Action Scenario.

Notes: The charts report differences between three-year cumulative impulse responses to a negative one-per-cent house price shock for different degrees of targeted intervention (ϵ) and their analogues in the no-policy-action scenario ($\epsilon = 0$). The same exercise is conducted for three levels of inefficiency costs (κ). This difference is defined as the three-year cumulative gains. Private debt is calibrated by setting m'' = 0.99 and m = 0.44. Y-axes for output are in percent deviations from steady state, while for private and public debt to GDP ratios they are in absolute deviations.

Two main results emerge from this exercise: (i) there is a non-zero level of targeted intervention, ϵ^* , that minimizes output losses and the increase in public debt to GDP following the shock;³⁰ (ii) the more efficient the intervention (the

 $[\]tilde{Y}_t = C_t + C'_t + C''_t + I_t + G_t.$

³⁰In the charts, these reduced losses are shown in terms of macroeconomic gains of intervening relative to not intervening.




Notes: The gain-maximizing degree of targeted intervention, ϵ^* , is computed as the level of ϵ that maximizes the difference between the three-year cumulative impulse response of output to a negative one-per-cent house price shock and its analogue in the no-policy-action scenario ($\epsilon = 0$). Private debt is calibrated by setting m'' = 0.99 and m = 0.44. The same exercise is conducted for two levels of inefficiency costs ($\kappa = 0$ and $\kappa = 0.1$).

lower the value of κ) the larger the benefits for output and public debt ratio of any level of targeted intervention and, accordingly, the higher the gain-maximizing level of targeted intervention (higher ϵ^*).

It is noteworthy that, when the level of targeted intervention is set equal to ϵ^* , the increase in the public debt-to-GDP ratio over three years is less than in the case of no government intervention. In fact, in the former case, the beneficial effect of targeted intervention on output exceeds the adverse impact of the subsequent fiscal consolidation required to neutralize the effect of additional public spending.³¹

In a further policy experiment we ask what the impact of targeted interventions

³¹In an exercise reported in the Fiscal Monitor (IMF, 2016), our DSGE model is used to assess the benefits of three alternative fiscal policy measures: targeted intervention, government consumption, and public investment. Results unveil that the output benefits of targeted intervention are four times larger than those of more standard stimulus measures. The rationale behind this result is that, by lending to credit-constrained households and firms, the government can leverage a much larger amount of spending than through other policy stimuli of equal cost. In fact, the fiscal cost of targeted intervention is only a fraction of the total government loan.

is when the government has modest fiscal space. To this end we simulate the model setting the value of the public debt-to-GDP ratio at higher steady-state levels, Γ . Figure 5 shows that, even when fiscal space is modest, targeted intervention remains beneficial for output in net terms, although the degree of intervention that minimizes output losses diminishes monotonically as fiscal space shrinks. This happens because higher government debt implies a dearer debt service, which demands a larger consolidation *ex post*. Inefficiency costs (the effect of which can be appreciated by setting κ to 0.1 from 0) worsen the picture because, other things equal, they augment the cost of the intervention, shifting the relationship between ϵ^* and Γ downward. This warrants an even lower degree of intervention. In fact, as shown in Figure E.1 (Appendix E), keeping the degree of targeted intervention fixed ($\epsilon = 0.1$), tax rates on consumption and labor income increase by more with higher steadystate levels of public debt-to-GDP to ensure fiscal solvency. The negative effects generated by the higher levels of (current and future) taxes depress consumption, labor, and hence output. The policy implication from this experiment is that, while more limited fiscal spaces harshen the policymaker's trade-offs, to a lesser extent, targeted intervention minimizes output losses also when fiscal buffers are smaller and public intermediation of funds is not completely efficient.

As a last exercise, we compute the threshold value of inefficiency costs, κ , above which intervention would become detrimental, i.e. it would produce a loss of output relative to the no-policy-action scenario. Figure 6 reports the gain-maximizing degree of targeted intervention, ϵ^* , as a function of κ for two levels of steady-state government debt ratios: $\bar{\Gamma}/4 = 0.71$, and $\bar{\Gamma}/4 = 0.92$, which capture baseline and "high" public debt cases as in the rest of the paper. As expected, in all cases, ϵ^* is a negative function of κ . The higher the inefficiency costs, the higher the costs of government intermediation of funds, the lower the gain maximizing degree of targeted intervention. In addition, the figure shows that higher levels of public debt cause a downward shift of the relationship, due to the larger sovereign risk premium and, hence, higher costs of intervention for any given level of κ . For a baseline level of the steady-state government debt ratio, ϵ^* is very small but positive even if the government is not able to recover any of the loans extended to the private sector, i.e. $\kappa = 1$. In contrast, with high public debt, ϵ^* turns zero if $\kappa \geq 0.3$.³²

³²In the absence of inefficiency costs ($\kappa = 0$) and sovereign risk premium, ϵ^* would be unbounded and the government could neutralize completely the deleveraging phase.





Notes: The gain-maximizing degree of targeted intervention, ϵ^* , is computed as the level of ϵ that maximizes the difference between the three-year cumulative impulse response of output to a negative one-per-cent house price shock and its analogue in the no-policy-action scenario ($\epsilon = 0$). Private debt is calibrated by setting m'' = 0.99 and m = 0.44. The same exercise is conducted for two levels of steady-state government debt ratios: $\overline{\Gamma}/4 = 0.71$ and $\overline{\Gamma}/4 = 0.92$, to capture baseline and high public debt cases.

6 Conclusion

Do the outstanding levels of private and public debt amplify swings in economic activity over the leverage cycle? Is government lending toward credit-constrained agents, i.e. targeted intervention, desirable at times of financial stress?

This paper attempts to answer these fundamental, and yet largely unanswered policy questions in the context of a general equilibrium model that stylizes private debt/public debt dynamics and can thus reproduce well the observed correlations between debt and future economic activity in the Euro Area. These imply that higher private debt leads deeper recessions, while higher public debt does not, unless the average level of public debt is especially high.

Our answer to the first question is yes, with some caveats. Our model predicts that economies with a larger stock of private debt tend to face more severe recessions following financial shocks. By contrast, the level of public debt has virtually no bearing on the severity of financial recessions, unless public debt is high to begin with. From this we arrive at the less obvious conclusion that hefty levels of private debt are as, or even more worrisome than hefty stocks of public debt for macroeconomic stability at times of heightened financial volatility.

Our answer to the second question is also yes. By alleviating the private sector's borrowing constraints, the government mitigates the recession and thus limits the drop in fiscal revenues associated with less favorable output realizations. However, two qualifications are in order. First, financial assistance during phases of financial deleveraging should not be confused with blanket fiscal stimuli: here we explore a specific targeted policy, i.e. lending to financially-constrained agents during phases of credit deleveraging, and not standard spending stimuli. Second, there is a clear trade-off between costs and benefits of intervention. This is because the economic costs of targeted intervention rise (i) with the level of public debt – as endogenous sovereign risk premia aggravate debt servicing and taxes need to increase by more, causing greater output losses; and (ii) with the inefficiency of public intervention in aid of financially-constrained agents. In line with these findings, simulations demonstrate the important role played by fiscal buffers, precisely because these allow greater degrees of targeted intervention.

Our findings also have implications for macro-financial surveillance, in suggesting that equal attention should be warranted towards risks posed by the evolution and levels of private indebtedness, relative to those traditionally believed to be associated with public indebtedness in isolation. In addition, limiting LTV ratios both in advanced and emerging market economies greatly reduces macro financial vulnerabilities associated with excessive credit booms and would limit the realizations of deeper and more prolonged recessions following episodes of financial instability.

References

- Adolfson, M., Laséen, S., Lindé, J., and Villani, M. (2007). Bayesian estimation of an open economy DSGE model with incomplete pass-through. *Journal of International Economics*, 72(2):481–511.
- Aghion, P., Howitt, P., and Mayer-Foulkes, D. (2005). The effect of financial development on convergence: Theory and evidence. *The Quarterly Journal of Economics*, 120(1):173–222.

Alesina, A., Carloni, D., and Lecce, G. (2012). The electoral consequences of large

fiscal adjustments. In *Fiscal policy after the financial crisis*, pages 531–570. University of Chicago Press.

- Alpanda, S. and Zubairy, S. (2017). Addressing household indebtedness: Monetary, fiscal or macroprudential policy? *European Economic Review*, 92:47–73.
- Angelini, P., Neri, S., and Panetta, F. (2014). The interaction between capital requirements and monetary policy. *Journal of Money, Credit and Banking*, 46(6):1073–1112.
- Arcand, J. L., Berkes, E., and Panizza, U. (2015). Too much finance? Journal of Economic Growth, 20(2):105–148.
- Arestis, P., Chortareas, G., and Magkonis, G. (2015). The financial development and growth nexus: A meta-analysis. *Journal of Economic Surveys*, 29(3):549–565.
- Austin, D. A. and Miller, R. S. (2011). Treasury securities and the US sovereign credit default swap market. In *Congressional Research Service*, CRS Report for Congress, number 7-5700.
- Azariadis, C. (2016). A Taylor rule for public debt. *Review*, 98(3):227–238.
- Baron, M. and Xiong, W. (2017). Credit expansion and neglected crash risk. The Quarterly Journal of Economics, 132(2):713–764.
- Bernanke, B., Gertler, M., and Gilchrist, S. (1999). The financial accelerator in a quantitative business cycle model. *Handbook of Macroeconomics*, 1:1341–1393.
- Bi, H. (2012). Sovereign default risk premia, fiscal limits, and fiscal policy. *European Economic Review*, 56(3):389–410.
- Bi, H. and Traum, N. (2014). Estimating fiscal limits: The case of Greece. *Journal* of Applied Econometrics, 29(7):1053–1072.
- Bohn, H. (1998). The behavior of US public debt and deficits. *the Quarterly Journal* of *Economics*, 113(3):949–963.
- Bohn, H. (2005). The sustainability of fiscal policy in the United States. Technical report.
- Calvo, G. (1983). Staggered prices in a utility-maximizing framework. Journal of monetary Economics, 12(3):383–398.
- Cantelmo, A. and Melina, G. (2018). Monetary policy and the relative price of durable goods. *Journal of Economic Dynamics and Control*, 86(C):1–48.

- Cantore, C., Levine, P., Melina, G., and Pearlmanl, J. (2018). Optimal fiscal and monetary policy, debt crisis and management. *Macroeconomic Dynamics*. Forthcoming.
- Card, D. (1991). Intertemporal labor supply: An assessment. NBER Working Papers 3602, National Bureau of Economic Research, Inc.
- Christensen, I., Corrigan, P., Mendicino, C., and Nishiyama, S.-I. (2016). Consumption, housing collateral and the Canadian business cycle. *Canadian Journal of Economics*, 49(1):207–236.
- Christiano, L., Motto, R., and Rostagno, M. (2010). Financial factors in economic fluctuations. *ECB Working Paper Series*, 1192.
- Clerc, L., Derviz, A., Mendicino, C., Moyen, S., Nikolov, K., Stracca, L., Suarez, J., and Vardoulakis, A. (2015). Capital regulation in a macroeconomic model with three layers of default. *International Journal of Central Banking*, 11(3):9–63.
- Commission, E. (2008). A European Economic Recovery Plan. Communication from the Commission to the European Council, COM(2008) 800.
- Corsetti, G., Kuester, K., Meier, A., and Müller, G. J. (2013). Sovereign risk, fiscal policy and macroeconomic stability. *Economic Journal*, 123(566):F99–F132.
- Dell'Ariccia, G., Igan, D., Laeven, L., Tong, H., Bakker, B., and Vandenbussche, J. (2012). Policies for macrofinancial stability: How to deal with credit booms. *IMF Staff Discussion Note*.
- Demirgüç-Kunt, A., Feyen, E., and Levine, R. (2013). The evolving importance of banks and securities markets. *The World Bank Economic Review*, 27(3):476–490.
- ECB (2009). Housing finance in the euro area. Technical report, European Central Bank, Structural Issues Report.
- ECB (2012). Corporate indebtedness in the euro area. Technical report, European Central Bank, Monthly Bulletin.
- Eggertsson, G. B. and Krugman, P. (2012). Debt, deleveraging, and the liquidity trap: A Fisher-Minsky-Koo approach. The Quarterly Journal of Economics, 127(3):1469–1513.
- Ergungor, O. E. (2008). Financial system structure and economic growth: Structure matters. *International Review of Economics & Finance*, 17(2):292–305.
- Farhi, E. and Werning, I. (2016). A theory of macroprudential policies in the presence of nominal rigidities. *Econometrica*, 84(5):1645–1704.

- Fecht, F., Huang, K. X., and Martin, A. (2008). Financial intermediaries, markets, and growth. *Journal of Money, Credit and Banking*, 40(4):701–720.
- Fisher, I. (1933). The debt-deflation theory of great depressions. *Econometrica*, 1(4):337–57.
- Forni, L., Monteforte, L., and Sessa, L. (2009). The general equilibrium effects of fiscal policy: Estimates for the euro area. *Journal of Public Economics*, 93(3-4):559–585.
- Geanakoplos, J., Axtell, R., Farmer, D. J., Howitt, P., Conlee, B., Goldstein, J., Hendrey, M., Palmer, N. M., and Yang, C.-Y. (2012). Getting at systemic risk via an agent-based model of the housing market. *The American Economic Review*, 102(3):53–58.
- Gerali, A., Neri, S., Sessa, L., and Signoretti, F. M. (2010). Credit and banking in a DSGE model of the euro area. Journal of Money, Credit and Banking, 42(s1):107– 141.
- Gertler, M. and Karadi, P. (2011). A model of unconventional monetary policy. Journal of Monetary Economics, 58(1):17–34.
- Ghosh, A. R., Kim, J. I., Mendoza, E. G., Ostry, J. D., and Qureshi, M. S. (2013). Fiscal fatigue, fiscal space and debt sustainability in advanced economies. *The Economic Journal*, 123(566).
- Glick, R. and Lansing, K. J. (2010). Global household leverage, house prices, and consumption. *FRBSF Economic Letter*.
- Gomes, S., Jacquinot, P., and Pisani, M. (2012). The EAGLE. a model for policy analysis of macroeconomic interdependence in the euro area. *Economic Modelling*, 29(5):1686–1714.
- Gomes, S., Jacquinot, P., and Pisani, M. (2016). Fiscal devaluation in the euro area: A model-based analysis. *Economic Modelling*, 52(PA):58–70.
- Guerrieri, L. and Iacoviello, M. (2017). Collateral constraints and macroeconomic asymmetries. *Journal of Monetary Economics*, 90:28–49.
- Guerrieri, V. and Lorenzoni, G. (2017). Credit crises, precautionary savings, and the liquidity trap. *The Quarterly Journal of Economics*, 132(3):1427–1467.
- Iacoviello, M. (2005). House prices, borrowing constraints, and monetary policy in the business cycle. American Economic Review, 95(3):739–764.
- Iacoviello, M. (2015). Financial business cycles. Review of Economic Dynamics, 18(1):140–164.

- Iacoviello, M. and Neri, S. (2010). Housing market spillovers: evidence from an estimated DSGE model. American Economic Journal: Macroeconomics, 2(2):125–64.
- IMF (2012). Dealing with household debt. In World Economic Outlook, pages 89–124.
- IMF (2016). Fiscal Monitor, October 2016: Debt: Use It Wisely. International Monetary Fund.
- Jordà, O., Schularick, M., and Taylor, A. M. (2013). When credit bites back. Journal of Money, Credit and Banking, 45(s2):3–28.
- Jordà, O., Schularick, M., and Taylor, A. M. (2014). Private credit and public debt in financial crises. *FRBSF Economic Letter*.
- Justiniano, A., Primiceri, G. E., and Tambalotti, A. (2015). Household leveraging and deleveraging. *Review of Economic Dynamics*, 18(1):3–20.
- Keane, M. and Rogerson, R. (2012). Micro and macro labor supply elasticities: A reassessment of conventional wisdom. *Journal of Economic Literature*, 50(2):464– 76.
- Kiyotaki, N. and Moore, J. (1997). Credit cycles. Journal of Political Economy, 105,2:211–248.
- Korinek, A. and Simsek, A. (2016). Liquidity trap and excessive leverage. The American Economic Review, 106(3):699–738.
- Lambertini, L., Nuguer, V., and Uysal, P. (2017). Mortgage default in an estimated model of the US housing market. Journal of Economic Dynamics and Control, 76:171–201.
- Levine, R. (1997). Financial development and economic growth: views and agenda. Journal of economic literature, 35(2):688–726.
- Levine, R., Loayza, N., and Beck, T. (2000). Financial intermediation and growth: Causality and causes. *Journal of Monetary Economics*, 46(1):31–77.
- Liu, Z., Wang, P., and Zha, T. (2013). Land-price dynamics and macroeconomic fluctuations. *Econometrica*, 81(3):1147–1184.
- Luintel, K. B., Khan, M., Arestis, P., and Theodoridis, K. (2008). Financial structure and economic growth. Journal of Development Economics, 86(1):181–200.
- Martin, P. and Philippon, T. (2017). Inspecting the Mechanism: Leverage and the Great Recession in the Eurozone. *American Economic Review*, 107(7):1904–1937.

- Mendoza, E. G. and Ostry, J. D. (2008). International evidence on fiscal solvency: Is fiscal policy "responsible"? *Journal of Monetary Economics*, 55(6):1081–1093.
- Mian, A. R. and Sufi, A. (2014). House of debt: how they (and you) caused the Great Recession and how we can prevent it from happening again. University of Chicago Press.
- Mian, A. R., Sufi, A., and Verner, E. (2017). Household debt and business cycles worldwide. *Quarterly Journal of Economics*, 132(4):1755–1817.
- Minsky, H. P. (1982). Can it happen again? Essays on Instability and Finance. Routledge.
- Ostry, J. D., Ghosh, A. R., and Espinoza, R. A. (2015). When should public debt be reduced? *IMF Staff Discussion Notes*, 15/10.
- Peia, O. and Roszbach, K. (2015). Finance and growth: time series evidence on causality. *Journal of Financial Stability*, 19:105–118.
- Schularick, M. and Taylor, A. M. (2012). Credit booms gone bust: Monetary policy, leverage cycles, and financial crises, 1870-2008. *American Economic Review*, 102(2):1029–61.
- Smets, F. and Villa, S. (2016). Slow recoveries: Any role for corporate leverage? Journal of Economic Dynamics and Control, 70(C):54–85.
- Smets, F. and Wouters, R. (2003). An estimated dynamic stochastic general equilibrium model of the euro area. Journal of the European Economic Association, 1(5):1123–1175.
- Taylor, A. M. (2012). Global financial stability and the lessons of history: A review of Carmen M. Reinhart and Kenneth S. Rogoff's this time is different: Eight centuries of financial folly. *Journal of Economic Literature*, 50(4):1092–1105.
- Taylor, J. B. (1993). Discretion versus policy rules in practice. Carnegie-Rochester Conference Series on Public Policy, 39(1):195–214.
- Villa, S. (2016). Financial frictions in the Euro Area and the United States: a Bayesian assessment. *Macroeconomic Dynamics*, 20(5):1313–1340.

Appendix

A Countries in Panel Regressions and Descriptive Statistics

	Private debt ($\%$ of GDP)		Public debt ($\%$ of GDP)		GDP)	
	Years	Average	Std. dev.	Years	Average	Std. dev.
Austria	1960-2014	92.29	38.36	1988-2014	68.27	9.28
Belgium	1970-2014	120.52	45.38	1980-2014	111.33	16.01
Finland	1970-2014	120.34	31.20	1980-2014	36.61	16.67
France	1969-2014	125.86	25.73	1980-2014	54.93	22.21
Germany	1960-2014	100.24	19.14	1991-2014	62.37	11.61
Greece	1970-2014	62.28	33.17	1980-2014	91.52	45.06
Ireland	1971 - 2014	135.17	85.22	1995 - 2014	61.43	33.97
Italy	1960-2014	79.55	21.70	1988-2014	109.02	11.61
Netherlands	1961-2014	141.13	70.35	1980-2014	63.21	10.56
Portugal	1960-2014	124.69	49.46	1990-2014	72.27	28.40
Spain	1970-2014	123.53	44.86	1980-2014	52.09	19.57

Table A.1: Countries in Panel Regressions and Descriptive Statistics

B Private Debt and GDP Growth at Different Frequencies

The growth literature has looked extensively at the effects of private debt on the macroeconomy. It is beyond the scope of the present work to provide an exhaustive review of these studies. As a background to our analysis, here we briefly summarize the three main relevant strands of the literature. The first finds that in the long term, higher private sector credit supports economic growth (e.g. Levine, 1997; Levine et al., 2000; Arestis et al., 2015, among many others). A second strand of the literature finds that not just the size but also the structure of the financial system matters for growth (Ergungor, 2008; Fecht et al., 2008; Luintel et al., 2008; Demirgüç-Kunt et al., 2013; Peia and Roszbach, 2015). The third strand of the literature examines the non-linear effect of financial development on growth. For example, Aghion et al. (2005) find that financial development – measured using private credit divided by GDP – has a positive but eventually vanishing effect

 Table B.1: Correlations Between Changes in Private Debt and Subsequent Real

 GDP Growth at Various Horizons

		-	-	"High" private debt
\overline{Corr}	$\left(\Delta_3\left(\frac{PRD}{Y}\right)_{t-1},\Delta_3 y_{t+3}\right)$	-0.321***	-0.436***	-0.338***
Corr	$\left(\Delta_5\left(\frac{PRD}{Y}\right)_{t-1},\Delta_5y_{t+5}\right)$	-0.318***	-0.389***	-0.383***
Corr	$\left(\Delta_{10}\left(\frac{PRD}{Y}\right)_{t-1},\Delta_{10}y_{t+10}\right)$	-0.077	0.010	-0.321***
Corr	$ \begin{pmatrix} \Delta_3 \left(\frac{PRD}{Y}\right)_{t-1}, \Delta_3 y_{t+3} \\ \Delta_5 \left(\frac{PRD}{Y}\right)_{t-1}, \Delta_5 y_{t+5} \end{pmatrix} \\ \begin{pmatrix} \Delta_{10} \left(\frac{PRD}{Y}\right)_{t-1}, \Delta_{10} y_{t+10} \\ \Delta_{15} \left(\frac{PRD}{Y}\right)_{t-1}, \Delta_{15} y_{t+15} \end{pmatrix} $	0.525***	0.498***	0.478***

Notes: $\frac{PRD}{Y}$ denotes the ratio of private debt to GDP, y denotes the log of real GDP, while $\Delta_k x_t = x_t - x_{t-k}$. The threshold value to distinguish low- and high-private-debt countries is the median of average the private-debt to-GDP ratio (120.52 percent of GDP). *, ** and *** denote significance at a 10, 5 and 1 percent level, respectively.

on steady-state GDP. Along this line, Arcand et al. (2015) show that there is a threshold above which finance (measured as private credit to GDP) starts having a negative effect on economic growth. The growth literature, however, is far from having reached a consensus. Differences among different studies include the choice of financial variable proxies, the kind of data used, as well as whether studies take the issue of endogeneity into account.

As already mentioned in the paper, at shorter frequencies – mostly 3 to 5 years – new empirical studies have shown that increases in private sector credit, especially household debt, may raise the likelihood of a financial crisis and can lead to lower future growth (e.g. Jordà et al., 2014; Mian et al., 2017, among others). This is because high indebtedness can cause significant debt overhang problems when a country faces negative shocks.

In Table B.1, we distinguish between high-frequency (3 to 5 years) and lowfrequency (10-15 years) correlations of changes in private debt with GDP growth. The table shows that the euro area data are consistent not only with the recent findings on the relationship between debt and growth at high frequencies that we examine in the paper, but also with those pointing to the existence of a positive finance-growth nexus in the long run. In particular, the negative association between changes of private debt and future growth at business cycle frequencies (3 to 5 years) is confirmed for both the full sample of euro area countries considered and for subsamples with "low" and "high" average private debt.³³ At a 10-year horizon,

³³The threshold value to distinguish "low"- and "high"-private-debt countries is the median of the average private-debt to-GDP ratio (120.52 percent of GDP). This criterion places Greece, Italy,

the picture is more heterogeneous as the correlation becomes approximately zero for the full sample and for countries with "low" private debt, while it remains negative for the "high"-debt countries, echoing to some extent the findings of the growth literature on nonlinear effects. At a 15-year horizon, the positive nexus between private debt and growth can be detected for all samples.

C Equilibrium conditions

C.1 Optimality conditions of households and entrepreneurs

Patient households

Intertemporal maximization yields the following first-order conditions with respect to C'_t , L'_t , B'_t , B^G_t and h'_t :

$$\mu'_t = \frac{1}{(1 + \tau_t^C) X_t'},\tag{C.1}$$

$$\left(1 - \tau_t^W\right) \frac{W_t'}{P_t} = \left(L_t'\right)^{\eta - 1} \left(1 + \tau_t^C\right) X_t',\tag{C.2}$$

$$\frac{1}{(1+\tau_t^C)X_t'} = \beta E_t \left[\frac{R_t}{(1+\tau_{t+1}^C)X_{t+1}'\Pi_{t+1}} \right],$$
(C.3)

$$\frac{1}{(1+\tau_t^C)X_t'} = \beta E_t \left[\frac{(1-\Delta_{t+1}^G)R_t^G}{(1+\tau_{t+1}^C)X_{t+1}'\Pi_{t+1}} \right],$$
(C.4)

$$\frac{q_t}{(1+\tau_t^C)X_t'} = \frac{\zeta e_t^H}{h_t'} + \beta E_t \left[\frac{q_{t+1}}{(1+\tau_{t+1}^C)X_{t+1}'} \right],$$
(C.5)

where μ'_t is the Lagrange multiplier associated to the budget constraint and $\Pi_{t+1} \equiv P_{t+1}/P_t$ represents the gross inflation rate. Equations (C.3) and (C.4) imply a non-arbitrage condition between the riskless interest rate and that on government bonds, whereby a sovereign risk spread arises, i.e. $R_t^G = E_t \left[\left(1 - \Delta_{t+1}^G \right)^{-1} \right] R_t$.

Austria, Germany, Finland and Belgium in the "low" private debt subsample; and Spain, Portugal, France, Netherlands and Ireland in the "high" private debt subsample.

Impatient households

Intertemporal maximization yields the following first-order conditions with respect to C''_t , L''_t , B''_t and h''_t :

$$\mu_t'' = \frac{1}{(1 + \tau_t^C) X_t''},\tag{C.6}$$

$$\left(1 - \tau_t^W\right) \frac{W_t''}{P_t} = \left(L_t''\right)^{\eta - 1} \left(1 + \tau_t^C\right) X_t'',\tag{C.7}$$

$$\frac{1}{(1+\tau_t^C)X_t''} = \beta'' E_t \left[\frac{R_t}{(1+\tau_{t+1}^C)X_{t+1}''\Pi_{t+1}}\right] + \lambda_t'' R_t,$$
(C.8)

$$\frac{q_t}{(1+\tau_t^C)X_t''} = \frac{\zeta e_t^H}{h_t''} + E_t \left[\frac{\beta'' q_{t+1}}{(1+\tau_{t+1}^C)X_t''} + \lambda_t'' m'' q_{t+1} \Pi_{t+1} \right], \quad (C.9)$$

where μ_t'' is the Lagrange multiplier associated to the flow of funds and λ_t'' is the Lagrange multiplier associated with the borrowing constraint.

Entrepreneurs

Maximization of function (11) subject to (9), (10), (12), (13), (14), (15) and the two quadratic adjustment costs yields the following first-order conditions with respect to $C_{e,t}$, $B_{e,t}$, $I_{e,t}$, $K_{e,t}$, $h_{e,t}$, $L'_{e,t}$, $L''_{e,t}$, and $P_{e,t}$ which, evaluated at the symmetric equilibrium, read as:

$$\mu_t = \frac{1}{(1 + \tau_t^C) X_t},\tag{C.10}$$

$$\mu_t = \lambda_t R_t + \gamma E_t \left[\mu_{t+1} \frac{R_t}{\Pi_{t+1}} \right], \tag{C.11}$$

$$\mu_t = \lambda_t R_t + \gamma E_t \left[\mu_{t+1} \frac{R_t}{\Pi_{t+1}} \right], \tag{C.12}$$

$$u_t = \mu_t \left[1 + \frac{\psi_K}{\delta} \left(\frac{I_t}{K_{t-1}} - \delta \right) \right], \tag{C.13}$$

$$u_{t} = \gamma E_{t} \left\{ \begin{array}{l} \mu_{t+1} \left[\frac{\psi_{K}}{\delta} \left(\frac{I_{t+1}}{K_{t}} - \delta \right) \frac{I_{t+1}}{K_{t}} - \frac{\psi_{K}}{2\delta} \left(\frac{I_{t+1}}{K_{t}} - \delta \right)^{2} \right] \\ + \left[\mu_{t+1} M C_{t+1} \frac{\omega_{Y_{t+1}}}{K_{t}} + (1 - \delta) u_{t+1} \right] \end{array} \right\},$$
(C.14)

$$\mu_t q_t = E_t \left\{ \gamma \mu_{t+1} \left[q_{t+1} + M C_{t+1} \frac{\nu Y_{t+1}}{h_t} \right] + m \lambda_t q_{t+1} \Pi_{t+1} \right\}, \quad (C.15)$$

$$w_t' = MC_t \frac{\alpha \left(1 - \omega - \nu\right) Y_t}{L_t'},\tag{C.16}$$

$$w_t'' = MC_t \frac{(1-\alpha)(1-\omega-\nu)Y_t}{L_t''},$$
(C.17)

$$0 = 1 + e_t^P \chi \left(M C_t - 1 \right) - \psi_P \left(\Pi_t - 1 \right) \Pi_t + \psi_P E_t \left[\gamma \frac{\mu_{t+1}}{\mu_t} \left(\Pi_{t+1} - 1 \right) \Pi_{t+1} \frac{Y_{t+1}}{Y_t} \right],$$
(C.18)

respectively, where λ_t is the Lagrange multiplier associated with the borrowing constraint, MC_t is the firm's marginal cost and u_t is Tobin's q.

C.2 Remaining equilibrium conditions

$$X_t' = C_t' - \theta C_{t-1}' \tag{C.19}$$

$$\left(1+\tau_t^C\right)C_t''+q_t\Delta h_t''+\frac{R_{t-1}B_{t-1}''}{\Pi_t}+\frac{R_{t-1}B_{g,t-1}''}{\Pi_t}=\left(1-\tau_t^W\right)w_t''L_t''+B_t''+B_{g,t}''$$
(C.20)

$$B_t'' = m'' E_t \left[\frac{q_{t+1} h_t'' \Pi_{t+1}}{R_t} \right]$$
(C.21)

$$X_t'' = C_t'' - \theta C_{t-1}''$$
(C.22)

$$Y_t + B_t + B_{g,t} = \left(1 + \tau_t^C\right) C_t + q_t \Delta h_t + \frac{R_{t-1}B_{t-1}}{\Pi_t} + \frac{R_{t-1}B_{g,t-1}}{\Pi_t} + w_t' L_t' + w_t'' L_t'' + I_t + \xi_{K,t} + \xi_{P,t}$$
(C.23)

$$R_t B_t = m E_t \left[q_{t+1} h_t \Pi_{t+1} \right]$$
 (C.24)

$$Y_{t} = e_{t}^{A} K_{t-1}^{\omega} h_{t-1}^{\nu} \left(L_{t}^{\prime} \right)^{\alpha(1-\omega-\nu)} \left(L_{t}^{\prime\prime} \right)^{(1-\alpha)(1-\omega-\nu)}$$
(C.25)

$$I_t = K_t - (1 - \delta) K_{t-1}$$
 (C.26)

$$\xi_{K,t} = \frac{\psi_K}{2\delta} \left(\frac{I_t}{K_{t-1}} - \delta \right)^2 K_{t-1} \tag{C.27}$$

$$\xi_{P,t} \equiv \frac{\psi_P}{2} \left(\frac{P_t}{P_{t-1}} - 1 \right)^2 Y_t \tag{C.28}$$

$$X_t = C_t - \theta C_{t-1} \tag{C.29}$$

$$Y_t = C_t + C'_t + C''_t + I_t + G_t (C.30)$$

$$h + h' + h'' = 1 (C.31)$$

$$\log\left(\frac{R_t}{R}\right) = \rho_r \log\left(\frac{R_{t-1}}{R}\right) + (1 - \rho_r) \left[\rho_\pi \log\left(\frac{\Pi_t}{\Pi}\right) + \rho_y \log\left(\frac{Y_t}{Y}\right)\right]$$
(C.32)

$$p_t^* = \frac{\exp(\eta_1 + \eta_2 \Gamma_t)}{1 + \exp(\eta_1 + \eta_2 \Gamma_t)}$$
(C.33)

$$\Gamma_t = \frac{B_t^G}{Y_t} \tag{C.34}$$

$$\Delta_t^G = p_t^* \bar{\Delta}^G \tag{C.35}$$

$$B_t^G = \left(1 - \Delta_t^G\right) \frac{R_{t-1}^G B_{t-1}^G}{\Pi_t} + G_t + \frac{\left(R_{t-1}^G - R_{t-1}\right) B_{t-1}^{int}}{\Pi_t} + \kappa B_t^{int} - T_t + \Xi_t \quad (C.36)$$

$$\Xi_{t} = \Delta_{t}^{G} \frac{R_{t-1}^{G} B_{t-1}^{G}}{\Pi_{t}} \tag{C.37}$$

$$T_t = \tau_t^C \left(C'_t + C''_t + C_t \right) + \tau_t^W \left(w'_t L'_t + w''_t L''_t \right) + \tau_t^L$$
(C.38)

$$B_t^{int} \equiv B_{g,t}'' + B_{g,t} \tag{C.39}$$

$$b_{g,t}^{\prime\prime} = -\epsilon b_t^{\prime\prime} \tag{C.40}$$

$$b_{g,t} = -\epsilon b_t \tag{C.41}$$

$$b_{g,t}'' = \frac{B_{g,t}'' - B_g''}{Y} \tag{C.42}$$

$$b_{g,t} = \frac{B_{g,t} - B_g}{Y} \tag{C.43}$$

$$b_t'' = \frac{B_t'' - B''}{Y}$$
(C.44)

$$b_t = \frac{B_t - B}{Y} \tag{C.45}$$

$$\log\left(\frac{\tau_t}{\tau}\right) = \rho \log\left(\frac{\tau_{t-1}}{\tau}\right) + (1-\rho) \left[e^{\phi \frac{B^G}{Y}} \rho_B \log\left(\frac{B^G_{t-1}}{B^G}\right)\right], \quad (C.46)$$

	Gain-maximizing degree of targeted intervention, ϵ^*				
	$\bar{\Gamma}/4 = 0.54$	$\overline{\Gamma}/4 = 0.71$	$\overline{\Gamma/4} = 0.92$		
Baseline	0.81	0.69	0.45		
$\eta = 2$	0.71	0.62	0.40		
$\delta = 0.035$	0.82	0.70	0.45		
$\rho_y = 0.2$	0.79	0.65	0.40		

Table D.1: Gain-Maximizing Degree of Targeted Intervention–Robustness Checks

Notes: The gain-maximizing degree of targeted intervention, ϵ^* , is computed as the level of ϵ that maximizes the difference between the three-year cumulative impulse response of output to a negative one-per-cent house price shock and its analogue in the no-policy-action scenario ($\epsilon = 0$). Private debt is calibrated by setting m'' = 0.99 and m = 0.44. we set $\overline{\Gamma}/4 = 0.54$, $\overline{\Gamma}/4 = 0.71$, $\overline{\Gamma}/4 = 0.92$ to capture low, average and high public debt cases, respectively as in the rest of the paper.

$$\log\left(\frac{G_t}{G}\right) = \rho \log\left(\frac{G_{t-1}}{G}\right) - (1-\rho) \left[e^{\phi \frac{B^G}{Y}} \rho_B \log\left(\frac{B^G_{t-1}}{B^G}\right)\right], \quad (C.47)$$

$$\tau_t^C = \tau_t \bar{\tau}^C \tag{C.48}$$

$$\tau_t^W = \tau_t \bar{\tau}^W \tag{C.49}$$

$$\tau_t^L = \tau_t \bar{\tau}^L \tag{C.50}$$

$$\log\left(\frac{e_t^{\varkappa}}{\bar{e}^{\varkappa}}\right) = \rho_{\varkappa} \log\left(\frac{e_{t-1}^{\varkappa}}{\bar{e}^{\varkappa}}\right) + \epsilon_t^{\varkappa}, \qquad \varkappa = \{A, H, P\}$$
(C.51)

D Robustness Checks to Selected Parameter Values

In Table D.1 we check if the gain-maximizing degree of targeted intervention, ϵ^* , is robust to a very different choice of the Frisch elasticity. We follow Adolfson et al. (2007) who, in an influential empirical paper on the euro area, use a value of the Frisch equal to 1, i.e. $1/(\eta - 1) = 1 \iff \eta = 2$. Under the alternative parametrization, the values of ϵ^* are fairly close to those obtained under the baseline, especially for a high level of public debt at the steady state. In a recent paper, Iacoviello (2015) sets a higher value of the capital depreciation rate: $\delta = 0.035$. Table D.1 shows that ϵ^* is virtually unaffected by this different choice of δ . Our final robustness check concerns the output response in the Taylor rule. In Table D.1 we examine how ϵ^* changes when using a double value for ρ_y , i.e. 0.2. It turns out that this different choice has only a very minor impact on the gain-maximizing degree of targeted intervention. Figure E.1: Three-Year Cumulative Responses to a Negative One-Per-Cent House Price Shock for Different Steady-State (SS) Public Debt/GDP Ratio in the Case of Targeted Intervention



Notes: Y-axes are in percent deviations from steady state. The degree of targeted intervention ϵ is equal to 0.1 and inefficiency costs κ are equal to 0.1.

E The Role of Fiscal Buffers in the Case of Targeted Intervention

This section highlights the transmission mechanism of the negative house price shock in the case of targeted intervention for different steady-state levels of the public debtto-GDP ratio. In this exercise we assume a moderate degree of targeted intervention and inefficiency costs, $\epsilon = 0.1$ and $\kappa = 0.1$ respectively, as in Subsection 5.3. Figure E.1 shows the three-year cumulative responses of output, consumption, tax rates and the sovereign risk premium to a negative house price shock for different steadystate public debt/GDP ratio.³⁴ When fiscal space is limited (i.e. for a high level of the public debt-to-GDP ratio), debt servicing costs increase due to the higher sovereign risk premium. As evident from equation (21) describing public debt dynamics, public debt increases by more, and this requires bolder fiscal consolidation as foreseen by the fiscal rules (equations 23 and 24). The resulting increase in the tax rates and the contraction of government spending depress consumption, labor, and hence output. This figure confirms the merits of having fiscal buffers for any given level of targeted intervention.

 $^{^{34}{\}rm Given}$ our calibration of fiscal rules the two tax rates move symmetrically from their respective steady states.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1158 Targeting policy-compliers with machine learning: an application to a tax rebate programme in Italy, by Monica Andini, Emanuele Ciani, Guido de Blasio, Alessio D'Ignazio and Viola Salvestrini (December 2017).
- N.1159 Banks' maturity transformation: risk, reward, and policy, by Pierluigi Bologna (December 2017).
- N. 1160 *Pairwise trading in the money market during the European sovereign debt crisis*, by Edoardo Rainone (December 2017).
- N.1161 Please in my back yard: the private and public benefitsof a new tram line in *Florence*, by Valeriia Budiakivska and Luca Casolaro (January 2018).
- N. 1162 *Real exchange rate misalignments in the euro area*, by by Michael Fidora, Claire Giordano and Martin Schmitz (January 2018).
- N. 1163 What will Brexit mean for the British and euro-area economies? A model-based assessment of trade regimes, by Massimiliano Pisani and Filippo Vergara Caffarelli (January 2018).
- N.1164 Are lenders using risk-based pricing in the consumer loan market? The effects of the 2008 crisis, by Silvia Magri (January 2018).
- N.1165 Listening to the buzz: social media sentiment and retail depositors' trust by Matteo Accornero and Mirko Moscatelli (February 2018)
- N. 1166 Banks' holdings of and trading in government bonds, by Michele Manna and Stefano Nobili (March 2018).
- N. 1167 Firms' and households' investment in Italy: the role of credit constraints and other macro factors, by Claire Giordano, Marco Marinucci and Andrea Silvestrini (March 2018).
- N.1168 *Credit supply and productivity growth*, by Francesco Manaresi and Nicola Pierri (March 2018).
- N. 1169 *Consumption volatility risk and the inversion of the yield curve*, by Adriana Grasso and Filippo Natoli (March 2018).
- N.1170 *The global component of inflation volatility*, by Andrea Carriero, Francesco Corsello and Massimiliano Marcellino (April 2018).
- N. 1171 *The potential of big housing data: an application to the Italian real-estate market*, by Michele Loberto, Andrea Luciani and Marco Pangallo (April 2018).
- N. 1172 ECB monetary policy and the euro exchange rate, by Martina Cecioni (April 2018).
- N. 1173 *Firms' investments during two crises*, by Antonio De Socio and Enrico Sette (April 2018).
- N. 1174 How can the government spending multiplier be small at the zero lower bound?, by Valerio Ercolani and João Valle e Azevedo (April 2018).
- N. 1175 Asset price volatility in EU-6 economies: how large is the role played by the ECB?, by Alessio Ciarlone and Andrea Colabella (June 2018).
- N. 1176 *Fixed rate versus adjustable rate mortgages: evidence from euro area banks*, by Ugo Albertazzi, Fulvia Fringuellotti and Steven Ongena (June 2018).
- N. 1177 Short term forecasts of economic activity: are fortnightly factors useful?, by Libero Monteforte and Valentina Raponi (June 2018).
- N. 1178 *Discretion and supplier selection in public procurement*, by Audinga Baltrunaite, Cristina Giorgiantonio, Sauro Mocetti and Tommaso Orlando (June 2018).
- N. 1179 *Labor market and financial shocks: a time varying analysis*, by Francesco Corsello and Valerio Nispi Landi (June 2018).
- N. 1180 On the unintended effects of public transfers: evidence from EU funding to Southern Italy, by Ilaria De Angelis, Guido de Blasio and Lucia Rizzica (June 2018).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBANESE G., G. DE BLASIO and P. SESTITO, *My parents taught me. evidence on the family transmission of values,* Journal of Population Economics, v. 29, 2, pp. 571-592, **TD No. 955 (March 2014).**
- ANDINI M. and G. DE BLASIO, Local development that money cannot buy: Italy's Contratti di Programma, Journal of Economic Geography, v. 16, 2, pp. 365-393, **TD No. 915 (June 2013).**
- BARONE G. and S. MOCETTI, *Inequality and trust: new evidence from panel data*, Economic Inquiry, v. 54, pp. 794-809, **TD No. 973 (October 2014).**
- BELTRATTI A., B. BORTOLOTTI and M. CACCAVAIO, Stock market efficiency in China: evidence from the split-share reform, Quarterly Review of Economics and Finance, v. 60, pp. 125-137, TD No. 969 (October 2014).
- BOLATTO S. and M. SBRACIA, *Deconstructing the gains from trade: selection of industries vs reallocation of workers*, Review of International Economics, v. 24, 2, pp. 344-363, **TD No. 1037 (November 2015).**
- BOLTON P., X. FREIXAS, L. GAMBACORTA and P. E. MISTRULLI, *Relationship and transaction lending in a crisis*, Review of Financial Studies, v. 29, 10, pp. 2643-2676, **TD No. 917 (July 2013).**
- BONACCORSI DI PATTI E. and E. SETTE, Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register, Journal of Financial Intermediation, v. 25, 1, pp. 54-76, TD No. 848 (February 2012).
- BORIN A. and M. MANCINI, Foreign direct investment and firm performance: an empirical analysis of *Italian firms*, Review of World Economics, v. 152, 4, pp. 705-732, **TD No. 1011 (June 2015).**
- BRAGOLI D., M. RIGON and F. ZANETTI, *Optimal inflation weights in the euro area*, International Journal of Central Banking, v. 12, 2, pp. 357-383, **TD No. 1045 (January 2016).**
- BRANDOLINI A. and E. VIVIANO, *Behind and beyond the (headcount) employment rate,* Journal of the Royal Statistical Society: Series A, v. 179, 3, pp. 657-681, **TD No. 965 (July 2015).**
- BRIPI F., *The role of regulation on entry: evidence from the Italian provinces*, World Bank Economic Review, v. 30, 2, pp. 383-411, **TD No. 932 (September 2013).**
- BRONZINI R. and P. PISELLI, *The impact of R&D subsidies on firm innovation*, Research Policy, v. 45, 2, pp. 442-457, **TD No. 960 (April 2014).**
- BURLON L. and M. VILALTA-BUFI, A new look at technical progress and early retirement, IZA Journal of Labor Policy, v. 5, **TD No. 963 (June 2014).**
- BUSETTI F. and M. CAIVANO, *The trend-cycle decomposition of output and the Phillips Curve: bayesian estimates for Italy and the Euro Area,* Empirical Economics, V. 50, 4, pp. 1565-1587, **TD No. 941** (November 2013).
- CAIVANO M. and A. HARVEY, *Time-series models with an EGB2 conditional distribution*, Journal of Time Series Analysis, v. 35, 6, pp. 558-571, **TD No. 947 (January 2014).**
- CALZA A. and A. ZAGHINI, *Shoe-leather costs in the euro area and the foreign demand for euro banknotes,* International Journal of Central Banking, v. 12, 1, pp. 231-246, **TD No. 1039 (December 2015).**
- CESARONI T. and R. DE SANTIS, *Current account "core-periphery dualism" in the EMU*, The World Economy, v. 39, 10, pp. 1514-1538, **TD No. 996 (December 2014).**
- CIANI E., *Retirement, Pension eligibility and home production,* Labour Economics, v. 38, pp. 106-120, **TD** No. 1056 (March 2016).
- CIARLONE A. and V. MICELI, Escaping financial crises? Macro evidence from sovereign wealth funds' investment behaviour, Emerging Markets Review, v. 27, 2, pp. 169-196, TD No. 972 (October 2014).
- CORNELI F. and E. TARANTINO, *Sovereign debt and reserves with liquidity and productivity crises*, Journal of International Money and Finance, v. 65, pp. 166-194, **TD No. 1012 (June 2015).**
- D'AURIZIO L. and D. DEPALO, An evaluation of the policies on repayment of government's trade debt in *Italy*, Italian Economic Journal, v. 2, 2, pp. 167-196, **TD No. 1061 (April 2016).**
- DE BLASIO G., G. MAGIO and C. MENON, Down and out in Italian towns: measuring the impact of economic downturns on crime, Economics Letters, 146, pp. 99-102, TD No. 925 (July 2013).
- DOTTORI D. and M. MANNA, *Strategy and tactics in public debt management*, Journal of Policy Modeling, v. 38, 1, pp. 1-25, **TD No. 1005 (March 2015).**

- LIBERATI D., M. MARINUCCI and G. M. TANZI, Science and technology parks in Italy: main features and analysis of their effects on hosted firms, Journal of Technology Transfer, v. 41, 4, pp. 694-729, TD No. 983 (November 2014).
- MARCELLINO M., M. PORQUEDDU and F. VENDITTI, *Short-Term GDP forecasting with a mixed frequency dynamic factor model with stochastic volatility,* Journal of Business & Economic Statistics, v. 34, 1, pp. 118-127, **TD No. 896 (January 2013).**
- RODANO G., N. SERRANO-VELARDE and E. TARANTINO, *Bankruptcy law and bank financing*, Journal of Financial Economics, v. 120, 2, pp. 363-382, **TD No. 1013 (June 2015).**
- ZINNA G., Price pressures on UK real rates: an empirical investigation, Review of Finance, v. 20, 4, pp. 1587-1630, **TD No. 968 (July 2014).**

2017

- ADAMOPOULOU A. and G.M. TANZI, Academic dropout and the great recession, Journal of Human Capital, V. 11, 1, pp. 35–71, **TD No. 970 (October 2014).**
- ALBERTAZZI U., M. BOTTERO and G. SENE, Information externalities in the credit market and the spell of credit rationing, Journal of Financial Intermediation, v. 30, pp. 61–70, TD No. 980 (November 2014).
- ALESSANDRI P. and H. MUMTAZ, *Financial indicators and density forecasts for US output and inflation*, Review of Economic Dynamics, v. 24, pp. 66-78, **TD No. 977 (November 2014).**
- BARBIERI G., C. ROSSETTI and P. SESTITO, *Teacher motivation and student learning*, Politica economica/Journal of Economic Policy, v. 33, 1, pp.59-72, **TD No. 761 (June 2010).**
- BENTIVOGLI C. and M. LITTERIO, Foreign ownership and performance: evidence from a panel of Italian firms, International Journal of the Economics of Business, v. 24, 3, pp. 251-273, **TD No. 1085** (October 2016).
- BRONZINI R. and A. D'IGNAZIO, *Bank internationalisation and firm exports: evidence from matched firmbank data*, Review of International Economics, v. 25, 3, pp. 476-499 TD No. 1055 (March 2016).
- BRUCHE M. and A. SEGURA, *Debt maturity and the liquidity of secondary debt markets*, Journal of Financial Economics, v. 124, 3, pp. 599-613, **TD No. 1049 (January 2016).**
- BURLON L., *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory,, v. 19, 4, pp. 789–810, **TD No. 961 (April 2014).**
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Macroeconomic effectiveness of non-standard monetary policy and early exit. a model-based evaluation, International Finance, v. 20, 2, pp.155-173, TD No. 1074 (July 2016).
- BUSETTI F., *Quantile aggregation of density forecasts,* Oxford Bulletin of Economics and Statistics, v. 79, 4, pp. 495-512, **TD No. 979 (November 2014).**
- CESARONI T. and S. IEZZI, *The predictive content of business survey indicators: evidence from SIGE,* Journal of Business Cycle Research, v.13, 1, pp 75–104, **TD No. 1031 (October 2015).**
- CONTI P., D. MARELLA and A. NERI, Statistical matching and uncertainty analysis in combining household income and expenditure data, Statistical Methods & Applications, v. 26, 3, pp 485–505, TD No. 1018 (July 2015).
- D'AMURI F., *Monitoring and disincentives in containing paid sick leave*, Labour Economics, v. 49, pp. 74-83, **TD No. 787 (January 2011).**
- D'AMURI F. and J. MARCUCCI, *The predictive power of google searches in forecasting unemployment,* International Journal of Forecasting, v. 33, 4, pp. 801-816, **TD No. 891 (November 2012).**
- DE BLASIO G. and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Journal of Regional Science, v. 57, 1, pp. 48-74, **TD No. 953 (March 2014).**
- DEL GIOVANE P., A. NOBILI and F. M. SIGNORETTI, Assessing the sources of credit supply tightening: was the sovereign debt crisis different from Lehman?, International Journal of Central Banking, v. 13, 2, pp. 197-234, TD No. 942 (November 2013).
- DEL PRETE S., M. PAGNINI, P. ROSSI and V. VACCA, Lending organization and credit supply during the 2008–2009 crisis, Economic Notes, v. 46, 2, pp. 207–236, TD No. 1108 (April 2017).
- DELLE MONACHE D. and I. PETRELLA, Adaptive models and heavy tails with an application to inflation forecasting, International Journal of Forecasting, v. 33, 2, pp. 482-501, TD No. 1052 (March 2016).

- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, v. 40, 10, pp. 2078-2096, **TD No. 877 (September 2012).**
- GIACOMELLI S. and C. MENON, Does weak contract enforcement affect firm size? Evidence from the neighbour's court, Journal of Economic Geography, v. 17, 6, pp. 1251-1282, TD No. 898 (January 2013).
- LOBERTO M. and C. PERRICONE, *Does trend inflation make a difference?*, Economic Modelling, v. 61, pp. 351–375, **TD No. 1033 (October 2015).**
- MANCINI A.L., C. MONFARDINI and S. PASQUA, *Is a good example the best sermon? Children's imitation of parental reading*, Review of Economics of the Household, v. 15, 3, pp 965–993, **D No. 958** (April 2014).
- MEEKS R., B. NELSON and P. ALESSANDRI, *Shadow banks and macroeconomic instability*, Journal of Money, Credit and Banking, v. 49, 7, pp. 1483–1516, **TD No. 939 (November 2013).**
- MICUCCI G. and P. ROSSI, *Debt restructuring and the role of banks' organizational structure and lending technologies*, Journal of Financial Services Research, v. 51, 3, pp 339–361, **TD No. 763 (June 2010).**
- MOCETTI S., M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Journal of Financial Services Research, v. 51, pp. 313-338, **TD No. 752 (March 2010).**
- MOCETTI S. and E. VIVIANO, *Looking behind mortgage delinquencies*, Journal of Banking & Finance, v. 75, pp. 53-63, **TD No. 999 (January 2015).**
- NOBILI A. and F. ZOLLINO, A structural model for the housing and credit market in Italy, Journal of Housing Economics, v. 36, pp. 73-87, **TD No. 887 (October 2012).**
- PALAZZO F., Search costs and the severity of adverse selection, Research in Economics, v. 71, 1, pp. 171-197, **TD No. 1073 (July 2016).**
- PATACCHINI E. and E. RAINONE, Social ties and the demand for financial services, Journal of Financial Services Research, v. 52, 1–2, pp 35–88, TD No. 1115 (June 2017).
- PATACCHINI E., E. RAINONE and Y. ZENOU, *Heterogeneous peer effects in education*, Journal of Economic Behavior & Organization, v. 134, pp. 190–227, **TD No. 1048 (January 2016).**
- SBRANA G., A. SILVESTRINI and F. VENDITTI, *Short-term inflation forecasting: the M.E.T.A. approach,* International Journal of Forecasting, v. 33, 4, pp. 1065-1081, **TD No. 1016 (June 2015).**
- SEGURA A. and J. SUAREZ, *How excessive is banks' maturity transformation?*, Review of Financial Studies, v. 30, 10, pp. 3538–3580, **TD No. 1065 (April 2016).**
- VACCA V., An unexpected crisis? Looking at pricing effectiveness of heterogeneous banks, Economic Notes, v. 46, 2, pp. 171–206, TD No. 814 (July 2011).
- VERGARA CAFFARELI F., One-way flow networks with decreasing returns to linking, Dynamic Games and Applications, v. 7, 2, pp. 323-345, **TD No. 734 (November 2009).**
- ZAGHINI A., A Tale of fragmentation: corporate funding in the euro-area bond market, International Review of Financial Analysis, v. 49, pp. 59-68, **TD No. 1104 (February 2017).**

2018

- ADAMOPOULOU A. and E. KAYA, *Young Adults living with their parents and the influence of peers*, Oxford Bulletin of Economics and Statistics, v. 80, pp. 689-713, **TD No. 1038 (November 2015).**
- BELOTTI F. and G. ILARDI Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, v. 202, 2, pp. 161-177, TD No. 1147 (October 2017).
- BRILLI Y. and M. TONELLO, Does increasing compulsory education reduce or displace adolescent crime? New evidence from administrative and victimization data, CESifo Economic Studies, v. 64, 1, pp. 15–4, TD No. 1008 (April 2015).
- BUONO I. and S. FORMAI *The heterogeneous response of domestic sales and exports to bank credit shocks,* Journal of International Economics, v. 113, pp. 55-73, **TD No. 1066 (March 2018).**
- CARTA F. and M. DE PHLIPPIS, You've Come a long way, baby. husbands' commuting time and family labour supply, Regional Science and Urban Economics, v. 69, pp. 25-37, TD No. 1003 (March 2015).
- CARTA F. and L. RIZZICA, *Early kindergarten, maternal labor supply and children's outcomes: evidence from Italy*, Journal of Public Economics, v. 158, pp. 79-102, **TD No. 1030 (October 2015).**

- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, v. 85, pp. 215-235, **TD No. 1077 (July 2016).**
- CECCHETTI S., F. NATOLI and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring*, International Journal of Central Banking, v. 14, 1, pp. 35-71, **TD No. 1025 (July 2015).**
- CIPRIANI M., A. GUARINO, G. GUAZZAROTTI, F. TAGLIATI and S. FISHER, *Informational contagion in the laboratory*, Review of Finance, v. 22, 3, pp. 877-904, **TD No. 1063 (April 2016).**
- NUCCI F. and M. RIGGI, *Labor force participation, wage rigidities, and inflation,* Journal of Macroeconomics, v. 55, 3 pp. 274-292, **TD No. 1054 (March 2016).**
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, v. 22, 2, pp. 661-697, TD No. 1100 (February 2017).

FORTHCOMING

- ALBANESE G., G. DE BLASIO and P. SESTITO, *Trust, risk and time preferences: evidence from survey data,* International Review of Economics, **TD No. 911 (April 2013).**
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, **TD No. 1098 (February 2017).**
- BARONE G., G. DE BLASIO and S. MOCETTI, *The real effects of credit crunch in the great recession: evidence from Italian provinces,* Regional Science and Urban Economics, **TD No. 1057 (March 2016).**
- BELOTTI F. and G. ILARDI, Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, **TD No. 1147 (October 2017).**
- BERTON F., S. MOCETTI, A. PRESBITERO and M. RICHIARDI, *Banks, firms, and jobs,* Review of Financial Studies, **TD No. 1097 (February 2017).**
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis*, Journal of the European Economic Association, **TD No. 909 (April 2013).**
- CIANI E. and C. DEIANA, *No Free lunch, buddy: housing transfers and informal care later in life*, Review of Economics of the Household, **TD No. 1117 (June 2017).**
- CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, **TD No. 985 (November 2014).**
- D'AMURI F., Monitoring and disincentives in containing paid sick leave, Labour Economics, TD No. 787 (January 2011).
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, **TD No. 1174 (April 2018).**
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, **TD No. 877 (September 2012).**
- GERALI A. and S. NERI, *Natural rates across the atlantic,* Journal of Macroeconomics, **TD No. 1140** (September 2017).
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, **TD No. 898 (January 2013).**
- LINARELLO A., Direct and indirect effects of trade liberalization: evidence from Chile, Journal of Development Economics, TD No. 994 (December 2014).
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, **TD No. 1025 (July 2015).**
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, **TD No. 871 (July 2012).**
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, TD No. 1100 (February 2017).